Abstract

A key feature of OECD economic growth since the early 1970s has been the secular decline in manufacturing's share of GDP and the secular rise of service sectors. This paper examines the role played by relative prices, technology, factor endowments, and labour market institutions in the process of 'de-industrialization.' We find a statistically significant and quantitatively important effect of levels of educational attainment. Furthermore, the production structure responds differently to the educational attainment of men and women. Finally, countries with stronger levels of employment protection are shown to adjust more slowly to changes in prices, technology, and factor endowments.

KEYWORDS: De-industrialization, Educational Attainment, Factor Endowments, Labour Market Institutions, Specialization JEL Classification: F0, J0, O3

This paper was produced as part of the Centre's Globalisation and Labour Markets Programmes

Acknowledgements

This paper is produced as part of the Labour Markets and Globalisation Programmes of the ESRC funded Centre for Economic Performance at the London School of Economics. The views expressed are those of the authors alone and should not be attributed to any institution to which the authors are affiliated. We are grateful to Charlie Bean, Francis Green, Susan Harkness, Tim Leunig, Tony Venables, Anna Vignoles, and seminar participants at the London School of Economics and the CEPR ERWIT Conference (Copenhagen) for helpful comments. We would also like to thank Damon Clark, Gunilla Dahlén, Simon Evenett, Maia Guella-Rotllan, Randy E. Ilg, Marco Manacorda, Steve McIntosh, Jan van Ours, Barbara Petrongolo, Glenda Quintini, Lupin Rahman, Toshiaki Tachibanaki, Ingrid Turtola, and Colin Webb for their help with the data. We are grateful to Martin Stewart for research assistance. All opinions, results, and any errors are the authors' responsibility alone.

Stephen Nickell is a member of the Monetary Policy Committee, Bank of England and the Centre for Economic Performance, London School of Economics. Contact: <u>s.j.nickel@lse.ac.uk</u>. Stephen Redding is a member of the Centre for Economic Performance and a lecturer in the Economics Department at the London School of Economics. Contact: <u>s.j.redding@lse.ac.uk</u>. Joanne Swaffield is a lecturer in the Department of Economics, University of York and a member of the Centre for Economic Performance, LSE. Contact: <u>jks4@york.ac.uk</u>.

Published by Centre for Economic Performance London School of Economics and Political Science Houghton Street London WC2A 2AE

© S. Nickell, S. Redding and J. Swaffield, submitted June 2002

ISBN 0753015668

Individual copy price: £5

Educational Attainment, Labour Market Institutions, and the Structure of Production

Stephen Nickell, Stephen Redding, Joanna Swaffield

September 2002

1.	Introduction	2
2.	Theoretical Framework	7
3.	Empirical Specification	10
4.	Data Description and Analysis	17
5.	Econometric Estimation	24
6.	Conclusions	33
Appendi	x A: Imperfect Competition and External Economies of Scale	35
Appendi	x B: Data Sources	36
Referenc	es	44

The Centre for Economic Performance is financed by the Economic and Social Research Council

1. Introduction

A key feature of economic growth in industrialized countries since the early 1970s has been the secular decline in manufacturing's share of GDP and the secular rise in the share of service sectors. Although these changes are common to all OECD countries, their magnitude and timing varies substantially. While the United Kingdom and United States were quick to 'deindustrialize', Germany and Japan have retained larger shares of manufacturing in GDP. A variety of explanations have been proposed for these changes, and there has been much popular debate concerning the causes and implications of de-industrialization.¹ However, there have been few systematic econometric analyses of the phenomenon, and what work there has been typically focuses on manufacturing to the exclusion of other sectors. This paper analyses the determinants of patterns of specialization across five broad industrial sectors (including Agriculture, Manufacturing, and Service industries) for 14 OECD countries since the mid-1970s. The analysis is firmly grounded in general equilibrium trade theory - our econometric equation is derived directly from the neoclassical theory of trade and production. Two popular explanations for deindustrialization are differential rates of technological progress and changes in relative prices. Our approach incorporates both of these considerations, while also allowing a role for factor endowments in explaining variation in the magnitude and timing of structural change and a role for labour market institutions in shaping the speed of such change.

A central theme of our analysis is that labour market outcomes and institutions play an important role in determining production structure. This theme emerges in a number of ways. First, one popular explanation for differences in industrial structure is levels of educational attainment. For example, Germany is frequently characterised as having high levels of the intermediate or vocational qualifications that are used intensively in the manufacturing sector.² We investigate this hypothesis using a new dataset on educational attainment constructed from individual-level information in labour force surveys.³ The use of labour force survey data means that we maintain a definition of educational attainment that is as consistent as possible across countries; our econometric analysis explicitly controls for any remaining cross-country variation in the classification of educational levels. Information is available on an annual basis, providing a considerable advance for time-series work over the data available every five years in Barro and

¹ See Owen (1999) and Turner (2001) for recent informal discussions.

² See, for example, Steedman and Wagner (1989), part of a wider body of case study and econometric work at the United Kingdom's National Institute of Economic and Social Research.

³ See Data Appendix for further information concerning the data used. Another paper exploiting individual-level labour force survey data is Machin and Van Reenen (1998).

Lee (1993), (2000).⁴ The period since the mid-1970s is characterized by increasing educational attainment in OECD countries, although the magnitude and timing of these changes again varies substantially across countries.

Second, the sample period is characterized by marked changes in female labour participation and education decisions. Between 1971 and 1981, female labour force participation rates in Canada rose from 39.9% to 51.8%, while the percentage of the female population with a college degree or equivalent rose from 4.9% (approximately 50% of the male level) in 1975 to 11.8% in 1994 (over 75% of the male level).⁵ It is plausible that male-female differences in labour market outcomes and their change over time are driven by largely exogenous cultural attitudes. Indeed, the extent of change varies markedly across countries with cultural attitudes. Thus, while the percentage of the female population with a college degree or equivalent in Japan rose rapidly between 1975 and 1994, it remained approximately 30% of the male level in 1994.

The variation in participation and education decisions between men and women is part of a wider literature that finds substantial differences in labour market outcomes between the two sexes. For example, a large number of papers find evidence of a substantial gender wage differential, even after controlling for observable characteristics (such as age, experience, and occupation) and unobserved heterogeneity.⁶ There is also substantial informal evidence that industry and occupation vary substantially between men and women. For example, in 1995 the percentage of men in total employment in the UK was 71.0% in manufacturing compared to 42.2% in Business Services and 50.0% in Other Services. Further variation is observed with levels of educational attainment. Thus, in 1995 the percentage of female employees with a college degree or equivalent in UK manufacturing was approximately 7% compared to 19% for men.⁷

However, despite much informal evidence of differences in occupation and industry of employment, there has been little attempt to systematically examine the implications of changes in female-male participation and education decisions for economies' production structures. This paper undertakes such an analysis for a broad cross-section of OECD countries. Have economies that have been particularly successful in attracting women into the labour market or particularly successful in increasing levels of female educational attainment tended to specialize in

⁴ The quality of the Barro and Lee (1993) data has recently been criticised by de la Fuente and Domenech (2000) and Krueger and Lindahl (2000). See also Cohen and Soto (2001).

⁵ The female labour force participation rates are taken from Killingsworth and Heckman (1986); see also Blundell and MaCurdy (2001). The educational attainment data are from Table 3 below.

⁶ Examples include Blinder (1973), Oaxaca (1973), Oaxaca and Ransom (1994), and Swaffield (1999). For a discussion of changes in the gender wage differential over time, see Desai *et al.* (1999).

a different set of industries from economies that have instead retained high levels of male participation and educational attainment? In order to separate out the effects of changes in labour market participation and changes in educational attainment, we exploit data on both the percentage of the labour force and of the working age population with particular levels of educational attainment. Our preferred measure is the number of men (women) out of the working age male (female) population with a particular level of educational attainment. This variable is pre-determined by the educational decisions of previous cohorts prior to becoming working age, and rises over time as later cohorts have chosen to acquire higher and higher levels of education. We are able to explicitly test whether the effect on production structure of endowments of a particular education level is the same for men and women.

Third, a large theoretical and empirical literature emphasizes the role of institutions and public policies in shaping labour market outcomes.⁸ In particular, work emphasizes the role of employment protection and job security provisions in determining the speed at which workers are reallocated from old and declining sectors to new and expanding ones. Thus, using firm-level data Hopenhayn and Rogerson (1993) find evidence of a negative effect of employment protection on aggregate productivity and growth through this mechanism. Using data on a cross-section of countries over time, Lazear (1990) finds statistically and quantitatively important effects on aggregate employment levels. In this paper, we directly examine the role of employment protection legislation in determining countries' speeds of adjustment to long-run changes in patterns of specialization. Our sample includes countries with very different extents of provision towards employment protection. The econometric equation for the share of a sector in a country's GDP that we estimate has an Equilibrium Correction Model (ECM) interpretation. Within this equation, we test econometrically whether employment protection affects the speed of adjustment towards long-run equilibrium.

The paper relates to four main strands of existing literature. First, there is a relatively informal economic history literature that has examined de-industrialization, often with particular emphasis on the United Kingdom and United States.⁹ With the beginning of the Industrial Revolution in the late-Eighteenth Century, the share of agriculture in UK GDP progressively declined from 18.4% in 1856 to 3.4% in 1964. The share of manufacturing in UK GDP continued to rise into the 1960s (from 22.2% in 1856 to 33.6% in 1964), with de-industrialisation

⁷ These figures on shares of industry employment by sex and by both sex and level of educational attainment are taken from Tables B2 and B3 in Appendix B, which are derived from individual-level information in the United Kingdom's New Earnings Survey.

⁸ See Nickell (1997) and Nickell and Layard (1999) for recent surveys.

⁹ See, for example, Crafts (1996), Kitson and Michie (1996) and Broadberry (1997).

emerging as a phenomenon in the 1970s.¹⁰ Two clear candidate explanations for deindustrialization that we consider are pervasive technological change and systematic changes in relative prices with economic development. These are forces that are to some extent common across countries, and can explain the shared experience of structural change across OECD countries. However, we also require an explanation for differences in the magnitude and timing of structural change. Here, country-specific changes in technology and relative prices are potential explanations, as well as countries' factor endowments.

Second, the paper relates to the empirical literature that estimates the relationship between factor endowments and the international location of production. Harrigan (1995) and Bernstein and Weinstein (1998) regress output levels on factor endowments in a specification derived directly from the *n*-good, *m*-factor Heckscher-Ohlin model.¹¹ Factor endowments are found to have a statistically significant and quantitatively important effect on levels of production. Hanson and Slaughter (1999) and Gandal, Hanson, and Slaughter (1999) examine the generalized Rybczynski Theorem using US state-level and Israeli data respectively. As predicted by the theory, they find that changes in output mix play an important role in absorbing changes in factor endowments.

The $n \times m$ Heckscher-Ohlin model is in fact a special case of neoclassical theory, which allows for cross-country differences in preferences and technology. Harrigan (1997) estimates the neoclassical model for 8 manufacturing industries across 10 OECD countries, and finds that levels of technology as well as factor endowments are an important determinant of patterns of specialization. Harrigan and Zakrajsek (2000) further investigate the relationship between factor endowments and production in manufacturing industries. Using data on a broad sample of developed and developing countries, factor endowments are found to be a major influence on specialization.

In contrast to these papers, we analyse the determinants of specialization in manufacturing and non-manufacturing industries. Particular emphasis is placed on the determinants of de-industrialization and the role of labour market outcomes and institutions in determining patterns of production. We exploit a newly constructed dataset on educational attainment, investigate the implications of distinguishing between labour market outcomes for men and women, and consider the role of labour market institutions in shaping the speed of adjustment to structural change.

¹⁰ These historical figures are from Matthews *et al.* (1982), Table 8.1.

¹¹ Another literature considers the relationship between factor endowments and international trade in factor services. See, for example, Leamer (1984), Bowen *et al.* (1987), Trefler (1995), Davis *et al.* (1997), Gabaix (1997), and Davis and Weinstein (1998a). For recent surveys of the empirical trade literature, see Leamer and Levinsohn (1995) and Helpman (1999).

Third, as discussed above, the paper relates to the large labour market literature on malefemale differentials and on the role of institutions in determining the pace of job creation and job destruction. However, while these ideas have been emphasized in the labour market literature, there has been very little analysis of their consequences in product markets. This paper seeks to bridge that gap by considering implications for countries' production structure and for the speed of adjustment to long-run changes in production structure.

Fourth, a recent literature has examined the relationship between international trade, skillbiased technological change, rising wage inequality, changes in relative unemployment rates, and changes in the relative supply of skilled and unskilled workers. The 1970s and 1980s were a period that saw substantial changes in the relative supply of skills as measured by either educational attainment or occupational status in many OECD countries. The same period was also characterised by a rise in the relative wage of skilled workers in the UK and US and changes in relative unemployment rates for most OECD countries (see, for example, the discussion in Katz and Murphy (1992), Nickell and Bell (1995), (1996), Berman *et al.* (1998), and Machin and Van Reenen (1998)).

Many studies have investigated the relative roles played by skill-biased technological change and international trade in explaining rising wage inequality and changes in relative unemployment rates.¹² However, with the exception of the work relating the location of production to factor endowments that was described above, there has been little attempt to examine the implications of changes in the relative supply of skills for OECD countries' patterns of production.¹³ This paper undertakes a systematic analysis of these implications for 14 OECD countries since the mid-1970s. We control for other determinants of production structure such as technology and relative prices, and pay careful attention to the role of labour market institutions and to the potential differences in labour market outcomes between men and women.

The paper is structured as follows. Section 2 introduces the theoretical framework, and derives a structural equation relating the share of a sector in a country's GDP to relative prices, technology, and factor endowments. Our main econometric specification comes directly from the theoretical model and is discussed in Section 3. Section 4 examines the raw data and discusses cross-country differences in both the magnitude and timing of structural change and in the evolution of educational attainment. Section 5 presents the main econometric results, and

¹² See, for example, Lawrence and Slaughter (1993), Wood (1994), Krugman (1995), Leamer (1998), Feenstra and Hanson (1999), and many others.

¹³ Maskus *et al.* (1994) and Maskus and Webster (1999) examine the implications of changing relative supplies of skills for the net factor content of trade of trade in the UK and US.

examines the contributions of the explanatory variables to observed structural change. Section 6 summarises our conclusions.

2. Theoretical Framework

The starting point for the analysis is the standard neoclassical theory of trade and production, as expounded by Dixit and Norman (1980) and Woodland (1982). Time is indexed by t, countries by $c \in \{1,...,C\}$, final goods by $j \in \{1,...,n\}$, and factors of production by $i \in \{1,...,m\}$. Each country is endowed with an exogenous vector v_{d} of factors of production. Within this framework, production is typically assumed to occur under conditions of perfect competition and subject to a constant returns to scale technology. However, as discussed in Appendix A, it is straightforward to incorporate imperfect competition, and external economies of scale may also be introduced in the production technology. We allow for differences in factor endowments across countries c and technology differences across both countries c and industries j.

General equilibrium in production may be represented using the revenue function $r_i(p_a, p_a)$, where p_a is a vector of industry output prices. Under the assumption that this function is twice continuously differentiable, the country's vector of profit-maximising net outputs, $y_i(p_a, v_a)$, is equal to the gradient of $r_i(p_a, v_a)$ with respect to p_a .¹⁴ The analysis allows for Hicks neutral and factor-augmenting technology differences. In our main specification, we consider Hicks-neutral technology differences across countries, industries, and time. In this case, the production technology takes the form $y_{q\bar{q}} = \theta_{q\bar{q}} F_j(v_{q\bar{q}})$, where $\theta_{q\bar{q}}$ measures technological efficiency in industry *j* of country *c* at time *t*. The revenue function is given by $r_c(p_a, v_a) = r(\theta_a p_a, v_a)$, where θ_a is an $n \times n$ diagonal matrix of the technology measures $\theta_{q\bar{q}}$.¹⁵ Changes in technology in industry *j* of country *c* have analogous effects on revenue to changes in the price of industry *j* output, and the economy's vector of net outputs continues to be given by the gradient of the revenue function with respect to p_a . If technology differences are instead modelled as factor-augmenting, the revenue function takes the form $r_c(p_a, v_a) = r(p_a, \delta_a v_a)$, where δ_a is an $m \times m$ diagonal matrix of the revenue function with respect to p_a . If the technology differences are instead modelled as factor-augmenting, the revenue function takes the form $r_c(p_a, v_a) = r(p_a, \delta_a v_a)$, where δ_a is an $m \times m$ diagonal matrix of factor quality differences.¹⁶ Again, the economy's vector of net outputs is the gradient of the revenue function with respect to p_a .

¹⁴ Formally, a sufficient condition for the revenue function to be twice continuously differentiable is that there are at least as many factors as goods: m > n. With n > m, production levels may be indeterminant, although this will depend on technology differences, trade costs, and whether or not there is joint production. The potential existence of production indeterminacy is really an empirical issue. We present empirical evidence that the specification below, including relative prices, technology, and factor endowments, provides a relatively successful model of patterns of production in OECD countries.

¹⁵ See Dixit and Norman (1980), pages 137-9.

¹⁶ See Dixit and Norman (1980).

Returning to the case of Hicks-neutral technology differences, we follow Harrigan (1997), Harrigan and Zakrajsek (2000), Kohli (1991), and Woodland (1982) in assuming a translog revenue function,¹⁷

$$\ln r(\theta, p, v) = \alpha_{00} + \sum_{j} \alpha_{0j} \ln \theta_{j} p_{j} + \frac{1}{2} \sum_{j} \sum_{k} \alpha_{jk} \ln(\theta_{j} p_{j}) . \ln(\theta_{k} p_{k})$$

$$+ \sum_{i} \beta_{0i} \ln v_{i} + \frac{1}{2} \sum_{i} \sum_{k} \beta_{ik} \ln v_{i} . \ln v_{k}$$

$$+ \sum_{j} \sum_{i} \gamma_{ji} \ln(\theta_{j} p_{j}) . \ln(v_{i})$$
(1)

where $j,k \in \{1,...,n\}$ index goods and $i,h \in \{1,...,m\}$ index factors. Symmetry of cross effects implies,

$$\alpha_{jk} = \alpha_{kj}$$
 and $\beta_{ih} = \beta_{hi}$ $\forall j, k, i, h$ (2)

Linear homogeneity of degree 1 in v and p requires,

$$\sum_{j} \alpha_{0j} = 1, \qquad \sum_{i} \beta_{0i} = 1, \qquad \sum_{j} \alpha_{kj} = 0$$
(3)
$$\sum_{j} \alpha_{kj} = 0 \qquad (3)$$

$$\sum_{i} \beta_{ih} = 0, \qquad \sum_{i} \gamma_{ji} = 0 \tag{4}$$

Differentiating the revenue function with respect to each p_j and imposing the linear homogeneity restrictions, we obtain the following equation for the share of industry *j* in country *c*'s GDP at time $t(s_{cit})$,

$$s_{cjt} = \alpha_{0j} + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{p_{ckt}}{p_{c1t}}\right) + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{\theta_{ckt}}{\theta_{c1t}}\right) + \sum_{i=2}^{M} \gamma_{ji} \ln\left(\frac{v_{cit}}{v_{c1t}}\right)$$
(5)

This equation provides a theory-consistent measure of a country's specialization in an industry (the share of the industry in the country's GDP), and relates that measure to three sets of underlying economic determinants: relative prices, technology, and factor endowments. The theoretical analysis allows for a large number of factors of production. In particular, labour may be differentiated along a wide variety of dimensions of skills, including levels of education, general skills, vectors of industry-specific skills, physical strength, analytical skills, communication and interpersonal skills, etc. In empirical work, labour endowments are typically disaggregated according to either occupation-based (non-production / production workers) or educational attainment-based measures of skills. We adopt an educational attainment-based measure of skills,

¹⁷ To save notation, country-time subscripts are suppressed except where important. See Christensen et al. (1973) for

but distinguish between male and female workers. This allows for the possibility that male and female workers of the same education level *have* or *are perceived to have* different vectors of these other characteristics or dimensions of skills.

A number of labour market studies present evidence that this is indeed the case. Using data from a survey of individual workers across a variety of industries and occupations in the United Kingdom, Ashton *et al.* (1998) find that female workers have above average levels of communication skills with both clients and their peers, but below average levels of manual and problem-solving skills. The reverse is true for male workers. Green and James (2001) provide independent survey evidence for the United Kingdom that male line managers tend to underestimate the skills of their female subordinates.¹⁸ Note that the coefficients on factor endowments in each industry regression correspond to *general equilibrium* effects. An actual or perceived difference in levels of skills relevant to only a few industries (eg physical strength in agriculture and mining; communication skills in service sectors) can result in different general equilibrium effects and hence different estimated coefficients on endowments of men and women of a given educational level.

The theoretical analysis underlying equation (5) assumes that factors of production are perfectly mobile across industries within a country. While this may be a reasonable approximation in the long-run, it is likely in practice to take time for factors of production to be reallocated from declining to expanding sectors. Equation (5) should therefore be interpreted as a long-run equilibrium relationship towards which the economy is evolving. In the empirical analysis below, we allow for a general process of dynamic adjustment towards this long-run relationship.

The translog revenue function implies coefficients on relative prices, technology, and factor endowments in equation (5) that are constant across industries and over time. This is true even without factor price equalization. Indeed, with cross-country differences in technology, factor price equalization will typically not be observed. The effect of cross-country differences in relative prices and technology on patterns of production is directly controlled for by the presence of the second and third terms on the right-hand side.

Finally, the analysis so far has made no assumptions about whether countries are large or small, and allows for both tradeable and non-tradeable goods. If countries are small and all goods are freely tradeable, the vector of relative prices will be determined exogenously on world markets. With either large countries or non-tradeable goods, relative prices will be endogenous

a related discussion of properties of the translog production technology.

and will depend in part on a country's factor endowments and levels of technology. In the econometric analysis that follows, we control for this endogeneity. For example, other things equal and on average across industries in a given country, the relative price of a non-traded good will be lower the more intensively it uses the country's relatively abundant factors of production.

3. Empirical Specification

Our main econometric equation is derived directly from the theoretical framework above. Augmenting the specification in equation (5) with a stochastic error, we obtain,

$$s_{cjt} = \alpha_{0j} + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{p_{ckt}}{p_{clt}}\right) + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{\theta_{ckt}}{\theta_{clt}}\right) + \sum_{i=2}^{M} \gamma_{ji} \ln\left(\frac{v_{cit}}{v_{clt}}\right) + \varepsilon_{cjt}$$
(6)

This relationship may be estimated separately for each industry *j*, pooling observations across countries *c* and time *t*. When imposing linear homogeneity, we normalise relative to manufacturing, so that manufacturing prices and technology are the excluded variables (manufacturing corresponds to industry 1 in equation (6)). Factor endowments are normalised relative to each country's endowment of physical capital. We consider a very general specification of the error term,

$$\varepsilon_{cjt} = \eta_{cj} + d_{jt} + \psi_{cjt} \tag{7}$$

where η_{ij} is a country-industry fixed effect, d_{ji} is a {0,1} industry-time dummy, and ψ_{iji} is a stochastic error.

The fixed effect controls for unobserved heterogeneity across countries in the determinants of patterns of international specialisation, which we allow to be correlated with the explanatory variables. For example, countries in which the share of manufacturing is high due to unobserved country characteristics (eg features of geography) may be precisely the countries characterised by a low relative price of the good etc. The industry-time dummies control for common macroeconomic shocks across countries in each industry.

Equation (6) is a static long-run equilibrium relationship between the share of a sector in GDP, relative prices, technology, and factor endowments. By construction, the share of sector j in GDP (s_{gi}) is bounded between 0 and 1, and is therefore I(0). However, in any finite sample, this variable may be I(1). This is particularly true of our sample period (1975-94), which is

¹⁸ For US evidence on sex differentials using matched employee-employer data, see Bayard, Hellerstein, Neumark, and Troske (1999). Neumark (1999) provides evidence on the importance of asymmetric information and employer perceptions in generating male-female wage differentials.

characterised by a secular decline in the share of agriculture and manufacturing in GDP and a secular rise in the share of Services. Similarly, relative prices, technology, and factor endowments may all be I(1). In this case, equation (6) should be interpreted as a long-run cointegrating relationship between the share of a sector in GDP and the right-hand side variables. In order for the equation to have this cointegrating interpretation, we require that the estimated residuals are I(0). In the empirical analysis below, we make use of recent advances in panel data cointegration techniques, and employ the panel data unit root tests of Levin and Lin (1992) and Maddala and Wu (1999).¹⁹

In practice, it may take time for resources to be reallocated from declining to expanding sectors. To allow for gradual adjustment towards long-run equilibrium, we also consider a dynamic specification, where equation (6) is augmented with the lagged dependent variable,

$$s_{cjt} = \delta_c s_{cjt-1} + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{p_{ckt}}{p_{c1t}}\right) + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{\theta_{ckt}}{\theta_{c1t}}\right) + \sum_{i=2}^{M} \gamma_{ji} \ln\left(\frac{v_{cit}}{v_{c1t}}\right) + \eta_{cj} + d_{jt} + \psi_{cjt}$$
(8)

where $(1-\delta)$ corresponds to the speed of adjustment.²⁰ This relationship has an equivalent Equilibrium Correction Model (ECM) representation,

$$\Delta s_{cjt} = (1 - \delta_c) \left[s_{cjt-1}^* - s_{cjt-1} \right] + (1 - \delta_c) \Delta s_{cjt}^* + \eta_{cj} + d_{jt} + \psi_{cjt}$$
(9)

where a superscript * denotes the long-run equilibrium value of a variable,

$$s_{cjt}^{*} = \frac{1}{1 - \delta_{c}} \left[\sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{p_{ckt}}{p_{clt}}\right) + \sum_{k=2}^{N} \alpha_{jk} \ln\left(\frac{\theta_{ckt}}{\theta_{clt}}\right) + \sum_{i=2}^{M} \gamma_{ji} \ln\left(\frac{v_{cit}}{v_{clt}}\right) \right]$$
(10)

The speed of adjustment towards long-run equilibrium is allowed to vary across countries with labour market institutions and policies that affect the reallocation of labour from declining to expanding sectors. We employ an OECD measure of the strength of employment protection,²¹ and include in our dynamic specification both the lagged dependent variable and the lagged dependent variable interacted with employment protection. That is, in terms of equations (8)-(10),

¹⁹ Other analyses of unit root tests and cointegration in a panel data context include Im *et al.* (1997), Pedroni (1999), Pesaran *et al.* (1998), and Quah (1994).

²⁰ The presence of the lagged dependent variable on the right-hand side in this specification implies that the fixed effects estimator will be biased. Nickell (1981) shows that this bias is asymptotically decreasing in the number of time periods (*T*), which in the present case is large ($T \cong 20$).

²¹ See Nickell (1997) for an analysis of the effects of this variable on equilibrium levels of unemployment.

$\delta_c = \delta_0 + \delta_1.emp_c$

where a positive and statistically significant value of δ_t implies that countries with greater levels of employment protection (*emp*) adjust more slowly towards long-run equilibrium.

To measure the price of industry *j* output in country $c(p_{gl})$ we use the producer price deflator. This is an index of industry *j* prices in country *c* at time *t* relative to their value in the same country in 1990, and takes the value 1 in 1990 in all countries. It provides information on changes in nominal prices in a particular country-industry over time. It does not capture the initial level of relative prices across countries and industries in 1990. However, this is controlled for by the country-industry fixed effect (η_{gl}) in equations (6) and (8). When imposing linear homogeneity, we normalise the price of industry *j* output in country $c(p_{gl})$ by the manufacturing price (p_{ell}). This yields a measure of the evolution of relative prices over time, with the initial level of relative prices again captured in the country-industry fixed effect (η_{gl}).

Note that, if all goods were freely tradeable so that goods prices were the same across countries ($p_{det} = p_{kt}$ for all *i*), the relative price term on the right-hand side of equations (6) and (8) could be replaced with a full set of industry-time dummies. More generally, if there are constant barriers to trade across countries in individual industries, the relative price term can be replaced by industry-time dummies and a country-industry fixed effect. Finally, if the reduction in trade barriers in an industry associated with multilateral trade liberalisation is common across countries, this can also be captured with industry-time dummies. Thus, with a country-industry fixed effect and time dummies included in the specification of the error term, an F-test of the statistical significance of the relative price variables provides a test of the null hypothesis that relative prices only differ across countries by a constant.

As discussed above, if countries are large and/or goods are non-tradeable, relative prices will be endogenous. This gives rise to a standard simultaneity problem. Other things equal, a positive shock to the share of an industry in a country's GDP will result in a lower relative price for that industry's output, imparting a downwards bias to the estimated coefficients on relative prices. Note that this bias operates in the opposite direction to the economic relationship that we are seeking to identify – a positive supply-side relationship between the relative price of a good and the share of the sector in a country's GDP. We address the simultaneity problem using Instrumental Variables estimation. Domestic relative prices are modelled as a function of a country's own factor endowments and technology and of world relative prices.

Actual data on the cost inclusive of freight (cif) price of imports are not available for each country in our sample. Therefore, we measure the local currency price of imports by constructing a relative price at the border series for each country and then converting this relative price into domestic prices using data on exchange rates and tariffs. The relative price at the border series for country *e* is constructed as a weighted sum of the relative prices of all other countries in the sample (expressed in a common currency (dollars)), where the weights are the shares of each of the other countries in country *e*'s imports. This captures the idea that prices will vary geographically with transport costs, and the price of imports will be more closely related to prices in neighbouring countries with whom much trade occurs than to prices in distant countries with whom little trade occurs. The first-stage regression for relative prices is thus,

$$\ln\left[\frac{p_{cjt}}{p_{clt}}\right] = \phi_{0j} + \sum_{i} \phi_{1ij} \ln v_{cit} + \sum_{k} \phi_{2kj} \ln \theta_{ckt} + \sum_{k} \phi_{3kj} \ln p_{ckt}^{B} + \phi_{4j} \ln e_{ct} + \phi_{5j} \ln \tau_{ct} + \omega_{cj} + D_{jt} + u_{cjt}$$
(12)

where p^{B} denotes price at the border, *e* is the nominal exchange rate, τ is the average tariff rate, ω_{ij} is a country-industry fixed effect, D_{ji} are industry-time dummies, and u_{iji} is a stochastic error.

Although the output of each of the industries in our sample is to some extent tradeable, the degree of tradability varies across industries (from, for example, agriculture to service sectors). The coefficients on prices at the border are therefore allowed to vary across industries, as are the coefficients on all the other exogenous variables in the first-stage regression. Note that the normalization relative to manufacturing when imposing linear homogeneity above means that we are concerned with variation in the price of each sector's output *relative to manufacturing*. Since the latter is itself highly tradeable, world prices should be an important determinant of domestic relative prices, as will be shown to be the case empirically below. Since factor endowments and technology also appear independently on the right-hand side of equations (6) and (8), the identification of price effects comes from variation in prices at the border, exchange rates, and tariffs. We show below that these are powerful instruments in the first-stage regression. The key identifying assumption is that these variables only affect the share of a sector in a country's GDP through relative prices. We test and provide evidence for this identifying assumption in the empirical analysis that follows.

To measure technology in sector *j* of country $c(\theta_{ij})$ we employ a superlative index number measure of Total Factor Productivity (TFP).²² In industry *j*, TFP in each country is evaluated relative to a common reference point – the geometric mean of all other countries in that industry. This is done in all years for that industry (e.g. we measure TFP in US Business Services in 1980 relative to the geometric mean of the Business Services industry across all countries in 1980) and for all industries *j*. The measure of relative TFP is given by,

$$\ln(RTFP_{cjt}) = \ln\left(\frac{Y_{cjt}}{\overline{Y}_{jt}}\right) - \overline{\sigma}_{cjt} \cdot \ln\left(\frac{L_{cjt}}{\overline{L}_{jt}}\right) - (1 - \overline{\sigma}_{cjt}) \cdot \ln\left(\frac{K_{cjt}}{\overline{K}_{jt}}\right)$$
(13)

where an upper bar above a variable denotes a geometric mean; Y is real value-added; L is labour input (hours worked); K is the real capital stock. The variable $\overline{\sigma}_{cjt} = 1/2.(\alpha_{cjt} + \overline{\alpha}_{jt})$ is the average of labour's share in value-added in country $c(\alpha_{cjt})$ and the geometric mean labour share $(\overline{\alpha}_{jt})$.

One problem in measuring TFP is that the observed share of labour in value-added is typically quite volatile. This suggests measurement error, and we therefore exploit the properties of the translog production function to smooth labour shares. Assuming a translog production technology, α_{gi} may be expressed as the following function of the capital-labour ratio and a country-industry constant,

$$\alpha_{cjt} = \xi_{cj} + \phi_j . \ln(K_{cjt} / L_{cjt})$$
(14)

If actual labour shares deviate from their true values by an i.i.d. measurement error term, the parameters of this equation can be estimated by fixed effects panel data estimation, where we allow the coefficient on the capital-labour ratio to vary across industries *j*. The fitted values from this equation are used as the labour cost shares in our calculation of relative TFP above.

It is well known that measured TFP tends to be pro-cyclical (see, for example, the discussion in Hall (1990)). Furthermore, TFP is actually constructed using data on real value-added. Positive shocks to industry *j* output that raise the share of the sector in GDP may therefore also raise measured TFP, giving rise to a spurious positive correlation between an industry's TFP and its share in the country's GDP. We wish to identify the long-run relationship between trend changes in technical efficiency and the shares of industries in GDP, abstracting

²² See, for example, Caves *et al.* (1982a), (1982b), Harrigan (1997), (1999), and Griffith, Redding, and Van Reenen (2000).

from such short-run shocks and business cycle fluctuations. We therefore follow the real business cycle literature (see, for example, Hodrick and Prescott (1997) and Baxter and King (1999)) in smoothing measured TFP using a Hodrick-Prescott filter. The filter separates the growth and cyclical components of the data, and we employ the standard accepted value for the smoothing parameter λ from the real business cycle literature (λ =1600 using quarterly data, which corresponds to λ =6.25 using annual data).²³

Our factor endowment variables include measures of physical capital, land area, and labour disaggregated by both level of educational attainment and sex. We make the standard assumption in the international trade literature that these country-level endowments are exogenous with respect to shocks to production structure. Our preferred measure of educational attainment is the number of men (women) out of the working age male (female) population with a particular level of educational attainment. This variable is pre-determined by the educational decisions of previous cohorts prior to becoming working age, and rises over time as later cohorts have chosen to acquire higher and higher levels of education.

Even within countries, there are differences in the evolution of male and female levels of educational attainment over time, and there are further differences between countries. We use this variation to separately identify the male and female coefficients, and to examine whether the effects of endowments of a given education level on production structure are the same for men and women. We also experiment with an alternative measure of educational attainment, defined as the number of either men or women out of the labour force with a particular level of educational attainment. This measure captures endowments of economically active individuals (whether employed or unemployed), and enables us to exploit exogenous variation in female labour force participation decisions over time.

The estimated coefficients on factor endowments in equations (6) and (8) are general equilibrium effects. Even if countries have identical technologies and preferences, the existence of many goods (n) and factors of production (m) means that we cannot make predictions about the effect of factor endowments on output of individual industries. Therefore, we should not necessarily expect a factor endowment to have a positive effect on the GDP share of one particular industry that uses that factor of production relatively intensively. Nevertheless, since with many goods and factors of production the theorems of the 2×2×2 Heckscher-Ohlin model

 $^{^{23}}$ The value of 6.25 for annual data is taken from Ravn and Uhlig (2001). They show analytically that this corresponds exactly to the standard accepted value of 1600 for quarterly data. Baxter and King (1999) propose a similar value of 10 for annual data. The results that follow are robust to the use of alternative values for the smoothing parameter.

hold in a weakened form as averages or correlations,²⁴ we should expect a pattern of estimated coefficients across all industries as a whole that is broadly consistent with information on factor intensity.

One final econometric concern is that our measures of relative prices, TFP, and factor endowments may be subject to measurement error. This may be a particular concern for the non-manufacturing sectors, and we therefore employ a number of controls for potential measurement error. First, TFP for each country-industry-time observation is measured relative to a common reference point (the industry-time geometric mean). Any error in measuring TFP that is common across countries for a particular industry-time observation will therefore be controlled for.

Second, when linear homogeneity is imposed in equations (6) and (8), TFP and prices in each country-industry are normalized by the manufacturing values for that country. Any error in measuring TFP or prices that is common across industries for a particular country-time observation will therefore also be controlled for. Third, the country-industry fixed effect included in the econometric estimation will control for any time-invariant errors of measurement for an industry in a particular country. It will also control, for example, for any remaining differences across countries in the classification of educational attainment levels. Fourth, the industry-time dummies included in the econometric estimation will capture any errors of measurement that are common across countries for a particular industry-time observation.

Finally, any remaining classical measurement error will attenuate the estimated parameters of interest towards zero, biasing the results away from the economic relationships that we seek to identify. The potential objection that measurement error may be greater for non-manufacturing sectors must also be counterbalanced against the fact that manufacturing is typically less than 30% of GDP in OECD countries, and there is a need to understand the remaining 70% of economic activity. The dataset that we employ is a later version of that already successfully used to analyse productivity convergence in manufacturing and non-manufacturing in two influential papers by Bernard and Jones (1996a), (1996b).²⁵

²⁴ See, for example, the discussion in Dixit and Norman (1980), Chapter 4.

²⁵ These papers also demonstrate the point that conclusions which hold for manufacturing may fail to hold in nonmanufacturing sectors. Bernard and Jones (1996a), (1996b) find that much of the observed convergence in aggregate productivity among OECD countries is driven by convergence in non-manufacturing sectors.

4. Data Description and Analysis

The main source of data in the empirical application is the OECD's International Sectoral Data Base (ISDB), which provides information for one-digit manufacturing and non-manufacturing industries on current price value-added, constant price value-added, employment, hours worked, and the stock of physical capital. Data on GDP and a country's aggregate endowment of physical capital are also obtained from the ISDB. Information on educational endowments comes from individual countries' labour force surveys, while data on arable land area are collected from the United Nations Food and Agricultural Organisation (FAO).²⁶

Our sample is an unbalanced panel of 14 OECD countries and 5 one-digit industries during the period 1975-94. The distribution of observations across countries and over time is given in Table 1A. Table 1B lists the 5 one-digit industries, together with a sixth industry 'Government Producers and Other Producers.' This is somewhat of a residual category, and is less likely to be characterized by profit-maximising behaviour. For these reasons, it is excluded in the econometric analysis that follows.²⁷ More detailed information on the disaggregated sectors included in each one-digit industry is given in Appendix B. Table 2 reports the evolution of industry shares of GDP over time in each of the 14 countries.

The sample period was characterised by a decline in the share of Agriculture in GDP in all countries, although the rate of decline varies substantially – from over 95% in Germany during 1975-93 to less than 30% in the Netherlands during 1975-94. In manufacturing, countries differ substantially in terms of the initial share of this industry in GDP: in Germany and Japan, manufacturing constituted about 30% of GDP in 1975, while, in Australia, Canada, and Denmark, it was responsible for only 20% of GDP.

All countries experienced a decline in manufacturing's share of GDP. However, the magnitude and timing of this decline varies across countries. In Australia and the United Kingdom, manufacturing's share of GDP declined by approximately 35% over the whole sample period, while, in Denmark and Finland, it fell by less than 10%. In the Netherlands and Norway, the decline was most rapid in the first half of the sample period, whereas, in Germany and Japan, most of the fall in manufacturing's share of GDP occurred in the second half of the sample period. In other countries, such as Italy and the United Kingdom, the rate of decline of manufacturing's share of GDP is roughly constant over time.

²⁶ See Appendix B for further information concerning the data sets used.

²⁷ Since the industry 'Government Producers and Other Producers' is excluded, there are no cross-industry restrictions on the estimated coefficients. The model is estimated separately for each industry, and therefore the inclusion of 'Government Producers and Other Producers' would only affect the results in the other industries to the extent that cross-industry restrictions were imposed.

Country	Period
1. Australia	1983-93
2. Belgium	1987-94
3. Canada	1976-92
4. Denmark	1984-92
5. Finland	1985-94
6. France	1983-92
7. West Germany	1985-93
8. Italy	1978-94
9. Japan	1976-94
10. Netherlands	1976-94
11. Norway	1976-91
12. Sweden	1976-94
13. United Kingdom	1976-93
14. United States	1976-93

Table 1A: Estimation sample by Country (observations on the 5 one-digit industries listed in Table 1B)

Table 1B: Industry Composition (International Standard Industrial
Classification (ISIC))

Industry	Industry	Further Details
	Code	
1. Agriculture	10	Agriculture, Hunting, Forestry and Fishing (ISIC 10)
2. Manufacturing	30	Manufacturing (ISIC 30)
3. Other Production	40	Mining and Quarrying (ISIC 20)
		Electricity, Gas, and Water (ISIC 40)
		Construction (ISIC 50)
4. Other Services	50	Wholesale and Retail Trade, Restaurants
		and Hotels (ISIC 60)
		Transport, Storage, and Communication (ISIC 70)
		Community, Social, and Personal Services (ISIC 90)
5. Business Services	60	Financial Institutions and Insurance (ISIC 82)
		Real Estate and Business Services (ISIC 83)
Excluded industry:	70	Producers of Government Services
Government/Other		Other Producers
Producers		

Notes: see Appendix B for detailed information on the disaggregated sectors included in each one-digit industry above.

Country	Year	Agric	Manuf	Other	Business	Other	Gov./
				Prod.	Services	Services	Other
Australia	1975	4.96	20.35	15.90	15.14	39.02	4.63
	1985	3.94	16.94	16.81	18.68	39.59	4.04
	1994	2.76	14.25	12.78	24.51	42.08	3.62
Belgium	1975	2.92	25.91	12.84	3.44 (a)	39.39	15.50
~	1985	2.32	22.67	10.30	5.76 (a)	44.01	14.94
	1994	1.57	19.70	9.75	5.45 (a)	50.09	13.44
Canada	1975	4.91	20.35	15.01	15.28	26.14	18.31
	1985	3.10	18.96	15.77	18.22	25.78	18.17
	1992	2.38	16.26	12.77	21.29	26.94	20.36
Denmark	1975	5.59	19.99	10.21	14.58	29.19	20.44
	1985	5.60	19.57	8.15	16.76	27.83	22.09
	1992	3.86	18.53	8.60	17.95	28.36	22.70
Finland	1975	10.54	26.05	14.06	12.69	21.38	15.28
	1985	8.06	25.09	10.92	14.37	22.95	18.61
	1994	5.47	24.42	8.13	18.82	22.23	20.93
France	1975	5.60	27.22	10.36	15.95	25.55	15.32
	1985	4.07	23.07	8.95	19.23	26.92	17.76
	1992	2.93	20.80	8.33	22.83	28.11	17.00
West	1975	2.88	35.40	10.10	4.64 (a)	26.28	14.32
Germany	1985	1.80	32.62	9.09	5.66 (a)	29.33	14.09
	1993	1.09	27.16	8.32	6.04 (a)	35.67	13.68
Italy	1975	7.14	27.43	13.59	5.11 (b)	35.48	11.25
•	1985	4.55	24.61	11.08	4.79 ^(b)	41.86	13.11
	1994	2.94	20.52	11.16	4.99 ^(b)	46.82	13.57
Japan	1975	5.28	29.05	11.74	12.93	16.73 (c)	10.05
	1985	3.06	28.37	10.98	14.78	20.22 (c)	9.74
	1994	2.05	23.49	13.25	17.17	22.21 (c)	9.64
Netherl.	1975	4.72	22.69	13.38	13.73	31.08	14.40
	1985	4.15	18.64	15.50	18.29	31.20	12.22
	1994	3.52	18.63	9.64	24.10	33.37	10.74
Norway	1975	5.01	21.81	12.21	14.32	30.99	15.66
-	1985	3.30	13.69	27.35	15.22	25.45	14.99
	1991	3.14	12.14	20.49	18.23	28.74	17.26
Sweden	1975	4.84	28.02	10.25	14.40	21.37	21.12
	1985	3.59	23.66	9.81	17.58	21.21	24.15
	1994	2.16	21.44	8.56	23.33	21.53	22.98
United	1975	2.58	28.21	11.33	15.71	24.73	17.44
Kingdom	1985	1.90	23.92	15.36	18.80	24.24	15.78
~	1993	1.88	19.94	9.76	24.54	28.69	15.19
United	1975	3.46	22.28	10.07	18.21	31.68	14.30
States	1985	2.07	19.47	10.78	23.08	31.85	12.75
	1993	1.65	17.39	8.08	26.74	33.05	13.09

Table 2: Shares of Industrial Sectors in a Country's GDP (per cent)^(a)

Notes: 'Government and Other Producers' (ISIC 70) is the excluded industry in the econometric analysis that follows. (a) Figures are for the sub-sector 'Financial Institutions and Insurance' (ISIC 82), and the numbers therefore sum to less than 100%. ^(b) Figures are for the sub-sector 'Financial Institutions and Insurance (ISIC 82). ^(c) Figures for 'Other Services' exclude the sub-sector 'Wholesale and Retail Trade, Restaurants and Hotels' (ISIC 60), and therefore the numbers sum to less than 100%. Source: OECD International Sectoral Database (ISDB). See Appendix B for more information concerning the data used.

The initial level of the share of Other Production in GDP varies from about 10% in Germany and the United States to over 15% in the natural resource rich countries of Australia and Canada. In all countries except Norway and Japan, the share of this sector in GDP declined during 1975-94. The share of Business Services in GDP rose in all countries for which data are available during 1975-94.²⁸ The increase was most rapid in Australia, the Netherlands, Sweden, and the United Kingdom, and was least rapid in Denmark and Norway. The share of Other Services in GDP rose in all countries except Denmark and Norway.

Table 3 reports male and female educational attainment as a percentage of the male and female population respectively for the years 1975, 1985, and 1994. Data for 1975 are only available for half of the 14 countries, and, therefore, the discussion here concentrates on the period 1985-94. As discussed earlier, this period was characterised by increases in educational attainment in the OECD. Considering men and women together, all countries in Table 3 experienced an increase in the share of the population with high education (college degree or equivalent).²⁹ The rate of increase varies markedly across countries: from 38% and 36% in Italy and the Netherlands during 1985-94, to 14% and 15% in Denmark and the United States during the same period.

Considering men and women separately, all countries in Table 3 display an increase in the share of the male population and the share of the female population with high education. The increase is typically largest for women, and this is reflected in a rise in the share of women in the total number of individuals (men plus women) with high education in all countries except France. There are notable differences in patterns of educational attainment across countries. Educational attainment in Germany and Norway is disproportionately concentrated in the medium education group relative to other OECD countries. In the United Kingdom in 1975, over 50% of the male population and over 60% of the female population were in the low education group, compared with less than 30% of the male population and less than 25% of the female population in the United States.

The share of the male population with high education typically exceeds the share of the female population. However, this is not always so - in France, Italy, and Sweden in 1985, the share of the female population with high education exceeded that of the male population. There are large changes in the relative educational levels of men and women over time, and the rate of change varies substantially across countries. In Australia in 1982 and Canada in 1975, the share

²⁸ For Belgium, Germany, and Italy, the data are for a sub-sector of Business Services: Financial Institutions and Insurance (ISIC 80).

of women with high education was just over half the value for men. However, by 1993 in Australia and 1994 in Canada, the share of women with high education was over 75% of the value for men. In contrast, in Japan, the share of women with high education in 1994 remained about 30% of the value for men.

Multiplying the percentage shares in Table 3 by the male and female population levels reported in Table 4, we obtain our preferred measure of countries' endowments of men and women with each education level. Table B1 of Appendix B shows that not only do educational endowments vary substantially across countries and over time, but there is also a large degree of variation in the intensity with which the 5 one-digit industries employ men and women of any given educational level.

Information on endowments of physical capital and arable land is also reported in Table 4. There is much variation in the relative abundance of these two factor endowments across countries. In 1975, the ratio of arable land to physical capital in the United States was approximately 10 times that in the Netherlands. Countries also display very different rates of physical capital accumulation, with the physical capital stock rising by 113% in Japan during 1975-92, compared with a rise of 34% in Denmark during the same period.

²⁹ The aggregate educational attainment data (for men and women together) are not reported in Table 3, but are a weighted average of those reported for men and women separately.

			Men			Women	
Country	Year	Low	Med	High	Low	Med	High
Australia	1982	.484	.438	.078	.623	.333	.044
	1985	.462	.448	.091	.592	.353	.055
	1993	.341	.542	.117	.441	.469	.090
Belgium	1975	-	-	-	-	-	-
	1986	.349	.600	.051	.457	.523	.021
	1994	.277	.649	.073	.350	.614	.035
Canada	1975	.272	.639	.089	.264	.687	.049
	1985	.198	.682	.120	.189	.727	.084
	1994	.127	.696	.147	.133	.723	.118
Denmark	1983	.337	.611	.053	.452	.531	.017
	1985	.240	.707	.053	.386	.595	.019
	1994	.190	.751	.055	.302	.665	.028
Finland	1984	.526	.387	.086	.562	.359	.079
	1985	.512	.399	.089	.547	.370	.082
	1994	.438	.440	.121	.440	.435	.125
France	1982	.469	.421	.109	.466	.405	.129
	1985	.425	.449	.125	.418	.436	.146
	1994	.307	.511	.181	.308	.485	.207
West Germany	1984	.148	.783	.109	.315	.647	.051
5	1985	.143	.774	.121	.311	.644	.057
	1994	.132	.758	.140	.252	.681	.083
Italy	1979	.529	.426	.044	.474	.477	.049
2	1985	.392	.550	.058	.330	.607	.063
	1994	.217	.715	.083	.172	.744	.094
Japan	1975	.433	.425	.142	.484	.487	.029
5 1	1985	.306	.501	.193	.336	.619	.045
	1994	.228	.534	.238	.226	.703	.071
Netherlands	1975	.371	.526	.103	.490	.460	.050
	1985	.216	.630	.153	.278	.621	.100
	1994	.146	.648	.206	.169	.671	.160
Norway	1976	.023	.862	.114	.018	.915	.067
,	1985	.030	.822	.147	.025	.869	.106
	1994	.029	.775	.195	.028	.803	.169
Sweden	1975	.553	.323	.123	.596	.301	.103
	1985	.410	.413	.177	.420	.398	.181
	1994	.293	.430	.203	.262	.443	.227
United Kingdom	1975	.514	.438	.048	.634	.352	.015
0	1985	.375	.528	.097	.447	.508	.046
	1994	.258	.618	.124	.314	.613	.073
United States	1975	.274	.549	.177	.229	.625	.146
	1985	.166	.598	.237	.122	.672	.206
	1994	.117	.605	.270	.083	.659	.257

Table 3: Education attainment as a Percentage of the Male and Female Working Age Populations (1975, 1985, and 1994)

Notes: educational attainment data are from individual-level information in country labour force surveys. Low corresponds to no education or primary education; Medium corresponds to secondary and/or vocational qualifications; High corresponds to college degree or equivalent. See Appendix B for further information concerning the data used.

Country	Year	Capital	Male Pop.	Female Pop.	Arable
Australia	1979	789.80	4777	4651	43932
	1985	971.04	5294	5148	47150
	1993	1226.92	5944	5828	46300
Belgium	1975	385.88	-	-	982
	1986	536.11	3338	3312	765
	1994	671.63	3378	3325	777
Canada	1975	1123.04	7649	7531	44000
	1985	1686.04	8946	8827	45900
	1992	2192.23	9756	9609	45370
Denmark	1983	374.89	1702	1673	2593
	1985	389.32	1716	1683	2601
	1992	441.74	1768	1721	2539
Finland	1984	353.87	1663	1663	2294
	1985	365.18	1672	1667	2276
	1994	457.47	1719	1685	2267
France	1982	2419.60	17674	17611	17651
	1985	2573.97	18181	18224	17923
	1992	3061.88	18797	18839	18046
West Germany	1984	3756.45	21259	21396	11952
,	1985	3845.38	21355	21385	11957
	1993	4716.85	28117	27127	11676
Italy	1977	2380.55	17800	18645	9359
5	1985	3054.87	19313	19973	9050
	1994	3911.39	19353	19607	8329
Japan	1975	2757.52	37180	38460	4460
5 1	1985	5276.81	40950	41360	4209
	1994	8572.96	43630	43360	3999
Netherlands	1975	583.50	4406	4322	759
	1985	732.83	5023	4899	826
	1994	886.90	5182	5353	885
Norway	1975	203.16	1266	1239	792
,	1985	319.33	1355	1314	858
	1991	376.38	1403	1355	892
Sweden	1975	414.12	2660	2599	3006
	1985	532.66	2729	2665	2922
	1994	669.49	2844	2754	2780
United	1975	1970.63	17554	17638	6883
Kingdom	1985	2464.50	18643	18555	7006
1 miguom	1993	3063.79	19019	18763	6081
United	1975	13658.82	68335	70560	186472
States	1985	18257.51	78450	80067	187765
	1993	22083.93	83768	84837	181950

Table 4: Endowments of Physical Capital (billions US dollars, 1990 prices), Population (thousands), and Arable Land Area (thousands of hectares)

Notes: capital is stock of real physical capital from OECD's International Sectoral Database (ISDB) (billions of 1990 US dollars). Male and Female Population data from individual country labour force surveys (thousands). Arable is arable land area from United Nations Food and Agricultural Organisation (FAO) (thousands of hectares). See Appendix B for further information concerning the data used.

5. Econometric Estimation

We begin by considering the static long-run relationship between the share of a sector in GDP, relative prices, technology, and factor endowments in equation (6). As reported in Table C1 of Appendix C, the majority of industry GDP shares and independent variables are I(1) during the sample period according to the Levin and Lin (1992) and Maddala and Wu (1999) panel data unit root tests. Columns (1)-(5) of Table 5 present the results of estimating equation (6) for Agriculture, Manufacturing, Other Production, Other Services, and Business Services using the within groups (fixed effects) estimator. As reported in the lower panel of the table, we reject the null hypothesis of a unit root in the residuals in the vast majority of cases with either the Levin and Lin or Maddala and Wu panel data tests. This finding that the residuals are I(0) provides support for the cointegrating interpretation given above.

The pattern of estimated coefficients on the relative price and TFP terms in Table 5 is consistent with the predictions of theory. In all 5 industries, the estimated own-industry price terms are positive and statistically significant at the 5% level (imposing linear homogeneity means that, in Manufacturing, the own-industry estimated coefficient is minus the sum of the estimated coefficients on the other industry terms). With the exception of Other Services, the estimated coefficients on the own-industry TFP terms are positive and statistically significant at conventional critical values (in Manufacturing, the own-industry effect is again minus the sum of the estimated effects on the other industry terms). We noted earlier that, if industry trade barriers differ across countries by a constant, the relative price terms can be replaced by a countryindustry fixed effect and industry-time dummies. However, an F-test of the null hypothesis that the estimated coefficients on the relative price terms are zero is rejected at the 5% level in all industries. This provides evidence of the importance of country-specific changes in industry relative prices over time.

Our measures of factor endowments have a statistically significant effect on patterns of production. The coefficients on individual endowments vary substantially across industries. For example, while endowments of low education men have a positive and statistically significant effect on the share of Agriculture and Other Services, the effect in Manufacturing and Business Services is negative with the Manufacturing coefficient significant at the 12% level. Male and female educational endowments have very different implications for patterns of production. For example, while endowments of medium education men have a negative and statistically significant effect on the share of business services in GDP, the effect of endowments of medium education women is positive and statistically significant. The null hypothesis that the estimated

24

coefficients on all male educational endowments equal those on female educational endowments is rejected at the 5% level in all industries except Manufacturing. This provides support for the idea that men and women of the same educational level have or are perceived to have different vectors of other characteristics or dimension of skills. It is consistent with the large labour market literature reviewed earlier that finds substantial differences between men and women in terms of labour market outcomes.

In Table 6, we examine the potential endogeneity of our measures of relative prices. The specification in equation (6) is re-estimated using Instrumental Variables. Our instruments for relative prices are: prices at the border, exchange rates, and average tariff rates (as described in more detail in Section 3). The instruments are highly statistically significant in the first stage-regression for relative prices. The lower panel of Table 6 reports *p*-values for an F-test of the null hypothesis that the coefficients on the excluded exogenous variables are equal to zero in the first-stage. In all cases, the null hypothesis is rejected at conventional critical values. The key identifying assumption in the IV estimation is that the instruments only affect the share of a sector in a country's GDP in so far as they influence domestic relative prices. The lower panel of Table 6 also reports the results of a Sargan test of the model's overidentifying restrictions. In all industries, we are unable to reject the null hypothesis that the excluded exogenous variables are uncorrelated with the residuals in the second-stage regression for GDP shares. This provides support for our key identifying assumption.

In general, the Instrumental Variables coefficients lie close to those estimated in Table 5 using within groups. The lower panel of Table 6 also reports the results of a Hausman test of the null hypothesis that within groups is consistent and efficient. In each industry, we are unable to reject the null hypothesis at conventional critical values. Taken together, this provides evidence that the within groups estimates are not substantially biased by the potential endogeneity of relative prices. Hence, in the remainder of the paper we focus on the within groups results.

25

GDP share _{cjt}	(1) Agriculture	(2) Manufact.	(3) Other Production	(4) Other Services	(5) Business Services
Obs	200	200	200	200	200
Years	1976-94	1976-94	1976-94	1976-94	1976-94
Mskill1 _{ct} (γ_1)	0.0355	-0.0290	0.0016	0.0358	-0.0129
	(5.55)	(-1.55)	(0.09)	(2.30)	(-1.07)
Mskill2 _{ct} (γ_2)	-0.0034	-0.0215	-0.0921	0.1356	-0.0809
	(-0.25)	(-0.83)	(-2.65)	(4.55)	(-3.72)
Mskill3 _{ct} (γ_3)	-0.0013	-0.0268	0.0249	0.0254	-0.0116
	(-0.19)	(-1.40)	(1.33)	(1.80)	(-0.74)
Fskill1 _{ct} (γ ₄)	-0.0378	0.0126	0.0150	-0.0290	-0.0003
	(-5.37)	(0.65)	(0.73)	(-1.76)	(-0.02)
Fskill2 _{ct} (γ ₅)	0.0036	0.0081	0.0420	-0.0383	0.0610
	(0.36)	(0.32)	(1.47)	(-1.60)	(3.35)
Fskill3 _{ct} (γ ₆)	0.0100	0.0070	-0.0448	-0.0007	0.0116
	(2.17)	(0.49)	(-3.10)	(-0.07)	(1.06)
Arable _{ct} (γ_7)	-0.0045	0.0204	0.0300	-0.1082	0.0140
	(-0.71)	(0.93)	(1.21)	(-4.76)	(1.12)
$P10_{ct}$ (α_1)	0.0275	0.0057	-0.0360	0.0138	0.0065
	(12.63)	(0.88)	(-2.97)	(1.87)	(1.16)
$P40_{ct}$ (α_2)	-0.0088	-0.0510	0.1277	-0.0241	-0.0262
	(-3.87)	(-7.61)	(10.22)	(-2.95)	(-5.75)
$P50_{ct}$ (α_3)	-0.0053	0.0101	-0.0343	0.0571	-0.0716
	(-1.07)	(0.66)	(-1.78)	(3.65)	(-7.30)
$P60_{ct}$ (α_4)	0.0046	-0.0446	-0.0091	-0.0481	0.1079
	(1.43)	(-3.35)	(-0.77)	(-3.99)	(13.65)
$\mathrm{TFP10}_{\mathrm{ct}}$ (β_1)	0.0220	-0.0058	0.0041	-0.0162	0.0103
	(8.58)	(-0.84)	(0.27)	(-1.64)	(1.49)
$\mathrm{TFP40}_{\mathrm{ct}}$ (β_2)	0.0029	-0.0374	0.0849	-0.0188	-0.0190
	(0.826)	(-4.36)	(4.97)	(-1.72)	(-2.21)
$\mathrm{TFP50}_{\mathrm{ct}}$ (β_3)	0.0066	0.0588	-0.0625	-0.0041	0.0018
\\ ··/	(1.32)	(3.06)	(-3.74)	(-0.33)	(0.19)
$\mathrm{TFP60}_{\mathrm{ct}}$ (β_4)	0.0088	-0.0421	0.0066	0.0432	0.0199
	(1.66)	(-2.00)	(0.38)	(2.98)	(1.77)
Year Dummies	yes	yes	yes	yes	yes
Country-Industry Fixed effect	yes	yes	yes	yes	yes
Maddala-Wu (p-value)	0.000	0.000	0.000	0.000	0.015
Levin-Lin (unit root t-statistic)	-8.311	-7.720	-4.462	-6.318	-6.163
Adjusted R ²	0.9764	0.9807	0.9654	0.9955	0.9945

Table 5: Baseline Within Groups Estimation

Notes: Asymptotic t-statistics in parentheses (based on Huber-White heteroscedasticity robust standard errors). Regressions are estimated industry-by-industry, pooling observations across countries and over time. Estimation is by within groups (least squares dummy variables). All specifications include country-industry fixed effects and full set of industry-year dummies. Dependent variable is share of a sector in a country's GDP. Independent variables: Mskill1 is log number of males out of working age male population with low education relative to stock of physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males are relative to physical capital; P10 is log price of industry 10 output (Agriculture) relative to industry 30 output (Manufacturing), and so on for the other industries; industry 40 is Other Production; industry 50 is Other Services; industry 60 is Business Services; TFP10 is a log superlative index number short-run fluctuations all TFP measures are smoothed with a Hodrick-Prescott filter; Maddala-Wu is the Maddala and Wu (1999) panel data test of the null hypothesis that there is a unit root in the residuals. Levin-Lin is the Levin and Lin (1992), Table 5: 1% critical value = -6.72; 5% critical value = -6.28; 10% critical value = -6.04.

GDP share _{cjt}	(1) Agriculture	(2) Manufact.	(3) Other	(4) Other	(5) Business
	-		Production	Services	Services
Obs	200	200	200	200	200
Years	1976-94	1976-94	1976-94	1976-94	1976-94
Mskill1 _{ct} (γ_1)	0.0453	-0.0269	0.0227	0.0257	-0.0100
	(5.08)	(-0.96)	(0.75)	(1.07)	(-0.65)
Mskill2 _{ct} (γ_2)	0.0034	-0.0223	-0.0823	0.1248	-0.0754
	(0.22)	(-0.52)	(-1.72)	(3.14)	(-3.12)
Mskill3 _{ct} (γ_3)	0.0021	-0.0068	0.0530	0.0015	-0.014
	(0.20)	(-0.25)	(1.87)	(0.07)	(-0.74)
Fskill1 _{ct} (γ ₄)	-0.0493	0.0066	-0.0041	-0.0182	-0.0070
	(-4.96)	(0.20)	(-0.12)	(-0.65)	(-0.41)
Fskill2 _{ct} (γ ₅)	-0.0026	-0.0258	0.0207	-0.0172	0.0550
	(-0.20)	(-0.53)	(0.58)	(-0.49)	(2.37)
$Fskill3_{ct}$ (γ_6)	0.0098	0.0220	-0.0456	0.0001	0.0139
	(1.712)	(0.96)	(-1.92)	(0.01)	(1.09)
$Arable_{ct}$ (γ_7)	-0.0158	0.0721	0.0562	-0.1465	0.0029
- (1)	(-0.28)	(1.38)	(0.97)	(-3.23)	(0.11)
P10 _{ct} (α_1)	0.0354	-0.0014	0.0294	-0.0279	-0.0118
	(5.23)	(-0.05)	(1.13)	(-1.37)	(-0.89)
$P40_{ct}$ (α_2)	-0.0092	-0.1055	0.0953	0.0239	-0.0289
	(-1.06)	(-2.50)	(2.70)	(0.89)	(-1.58)
$P50_{ct}$ (α_3)	-0.0234	0.1222	0.0388	-0.0449	-0.0924
	(-1.35)	(1.43)	(0.50)	(-0.75)	(-2.30)
$P60_{ct}$ (α_4)	0.0256	0.0123	0.0019	-0.0396	0.1335
	(1.52)	(0.24)	(0.04)	(-0.93)	(4.18)
TFP10 _{ct} (β_1)	0.0206	-0.0177	0.0329	-0.0372	-0.0045
	(4.16)	(-0.96)	(1.36)	(-2.23)	(-0.37)
TFP40 _{ct} (β_2)	0.0041	-0.0582	0.0668	0.0002	-0.0152
- ct (1-2)	(0.81)	(-2.59)	(2.06)	(0.01)	(-1.11)
$\mathrm{TFP50}_{\mathrm{ct}}$ (β_3)	-0.0018	0.1274	0.0013	-0.0780	-0.0156
et - (j- 3)	(-0.16)	(2.18)	(0.03)	(-2.22)	(-0.62)
$TFP60_{ct}$ (β_4)	0.0252	0.0262	0.0165	0.0421	0.0450
	(1.45)	(0.48)	(0.28)	(0.95)	(1.30)
Year Dummies	yes	yes	yes	yes	yes
Country-Industry Fixed Effect	yes	yes	yes	yes	yes
First-Stage F-test (p-value)	0.000	-	0.000	0.000	0.001
Hausman <i>(p-value)</i>	0.999	0.999	0.999	0.999	0.999
Sargan (p-value)	0.978	0.788	0.879	0.986	0.999
Adjusted R^2	0.9695	0.9604	0.9407	0.9909	0.9938

Table 6: Instrumental Variables Estimation

Notes: Asymptotic t-statistics in parentheses (based on Huber-White heteroscedasticity robust standard errors). Regressions are estimated industry-by-industry, pooling observations across countries and over time. Estimation is by Instrumental Variables. All specifications include country-industry fixed effects and full set of industry-year dummies. Dependent variable is share of a sector in a country's GDP. Independent variables: Mskill1 is log number of males out of working age male population with low education relative to stock of physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with migh education relative to physical capital; Fskill1-Fskill3 are female educational endowments defined analogously; Arable is log arable land area relative to physical capital; P10 is log price of industry 10 output (Agriculture) relative to industry 30 output (Manufacturing), and so on for the other industries; industry 40 is Other Production; industry 50 is Other Services; industry 60 is Business Services; TFP10 is a log superlative index number measure of TFP in industry 10 relative to TFP in industry 30, and so on for the other industries. To abstract from short-run fluctuations all TFP measures are smoothed with a Hodrick-Prescott filter. Endogenous variables: P10, P40, P50, and P60. Exogenous variables Mskill1-Mskill3, Fskill1-Fskill3, TFP10-TFP60, Z10-Z60, Exrate, and Tariff. Z10 is a measure of log price at the border, calculated as trade-weighted sum of the price of industry 10 output in all other countries; Exrate is log nominal exchange rate; Tariff is the log average tariff rate. First-stage F-test is a test of the joint significance of the excluded exogenous variables in the first-stage r

In practice, it is likely to take time for factors of production to be reallocated from declining to expanding sectors. Equation (6) should therefore be interpreted as a long-run relationship towards which gradual adjustment occurs. The speed at which factors of production are able to be reallocated may depend on labour market policies and institutions – in particular, on employment protection provisions that limit the ability of firms in declining sectors to shed labour or raise the cost of them doing so. Table 7 therefore augments equation (6) with a lagged dependent variable and with the lagged dependent variable interacted with a measure of the extent of employment protection.³⁰ We thus arrive at the dynamic specification in equation (8), where the speed of adjustment towards long-run equilibrium is given by $(1-\delta_0-(\delta_1\times EmProt_c)))$. For convenience, the employment protection measure is normalized by its mean across countries; this normalization implies that one minus the estimated coefficient on the lagged dependent variable $(1-\delta_0)$ can be interpreted as a measure of the speed of adjustment at sample mean values of employment protection.

The estimated coefficient on the lagged dependent variable in Table 7 is positive and highly statistically significant in all industries, providing evidence of partial adjustment. The employment protection interaction is positively signed and statistically significant in the 3 industries that declined as a share of GDP during the sample period – Agriculture, Manufacturing, and Other Production. This provides evidence that countries with higher levels of employment protection were slower to reallocate resources away from these sectors in response to a change in long-run patterns of specialization. The employment interaction is positively signed although not statistically significant in the 2 industries which expanded as a share of GDP during the sample period – Other Services and Business Services. This is consistent with the idea that the main effect of employment protection is to raise the cost of shedding labour in declining sectors. The effects of employment protection are not only statistically significant but also quantitatively important. Moving from the country with the lowest levels of employment protection (the United States) to the country with the highest (Italy) raises the estimated coefficient on the lagged dependent variable in Manufacturing from 0.44 to 0.78 (an increase of over 75%) and reduces the implied speed of adjustment from 0.56 to 0.22.

³⁰ See Appendix A and Nickell (1997) for further information concerning the employment protection measure.

GDP share _{ct}	(1) Agriculture	(2) Manufact.	(3) Other	(4) Other	(5) Business
	0		Production	Services	Services
Obs	200	200	200	200	200
Years	1976-94	1976-94	1976-94	1976-94	1976-94
GDP share _{ct-1} (δ_0)	0.4297	0.5885	0.4452	0.6229	0.6247
	(4.810)	(10.224)	(3.455)	(7.070)	(11.065)
EmProt*GDP share _{ct-1} (δ_1)	0.0166	0.0181	0.0353	0.0139	0.0068
	(2.164)	(2.760)	(2.546)	(1.380)	(0.928)
Mskill1 _{ct} (γ_1)	0.0236	-0.0362	0.0354	0.0058	0.0028
	(3.155)	(-2.565)	(2.217)	(0.465)	(0.288)
Mskill2 _{ct} (γ_2)	0.0091	0.0159	-0.0338	0.0468	-0.0581
	(0.760)	(0.801)	(-1.889)	(2.574)	(-4.097)
Mskill 3_{ct} (γ_3)	0.0019	-0.0096	0.0212	0.0054	0.0015
	(0.323)	(-0.630)	(1.612)	(0.593)	(0.142)
Fskill1 _{ct} (γ ₄)	-0.0242	0.0280	-0.0394	0.0043	-0.0029
	(-2.847)	(1.790)	(-2.159)	(0.295)	(-0.253)
Fskill2 _{ct} (γ_5)	-0.0047	-0.0164	0.0227	-0.0087	0.0394
	(-0.504)	(-0.942)	(1.392)	(-0.572)	(2.973)
Fskill3 _{ct} (γ_6)	0.0029	-0.0036	-0.0344	-0.0028	0.0049
	(0.723)	(-0.302)	(-3.589)	(0.393)	(0.602)
Arable _{ct} (γ_7)	0.0025	0.0576	0.0052	-0.0587	-0.0004
	(0.443)	(3.387)	(0.363)	(-4.699)	(-0.044)
$P10_{ct}$ (α_1)	0.0196	-0.0018	-0.0294	0.0156	0.0031
	(7.943)	(-0.357)	(-2.686)	(2.625)	(0.764)
$P40_{ct}$ (α_2)	-0.0042	-0.0258	0.0669	-0.0138	-0.0081
1 10tt (002)	(-2.119)	(-5.061)	(3.943)	(-2.497)	(-2.364)
$P50_{ct}$ (α_3)	-0.0002	0.0198	-0.0160	0.0175	-0.0319
	(-0.037)	(1.936)	(-1.078)	(1.574)	(-3.867)
$P60_{ct}$ (α_4)	-0.0025	-0.0321	-0.0060	-0.0233	0.0559
10000 (004)	(-0.825)	(-3.438)	(-0.677)	(-2.579)	(7.561)
TFP10 _{ct} (β_1)	0.0106	-0.0239	0.0016	0.0063	0.0020
	(3.524)	(-3.219)	(0.170)	(0.943)	(0.355)
TFP40 _{ct} (β_2)	0.0022	-0.0130	0.0656	-0.0102	-0.0170
(12)	(0.625)	(-1.759)	(3.733)	(-1.075)	(-2.618)
TFP50 _{ct} (β_3)	0.0050	0.0316	-0.0396	0.0128	-0.0076
- ~u (F.)/	(1.294)	(2.802)	(-3.003)	(1.309)	(-0.913)
$\mathrm{TFP60}_{\mathrm{ct}}$ (β_4)	-0.0004	-0.0357	0.0204	-0.0006	0.0235
···· (P+)	(-0.097)	(-2.919)	(1.647)	(-0.063)	(2.808)
Year Dummies	yes	yes	yes	yes	yes
Country-industry Fixed Effect	yes	yes	yes	yes	yes
Adjusted R ²	0.9830	0.989	0.979	0.998	0.997

 Table 7: Dynamic Within Groups Estimation: Partial Adjustment and the Role of Employment Protection

Notes: asymptotic t-statistics in parentheses (based on Huber-White heteroscedasticity robust standard errors). Regressions are estimated industry-by-industry, pooling observations across countries and over time. Estimation is by within groups (least squares dummy variables). All specifications include country-industry fixed effects and full set of industry-year dummies. Dependent variable is share of a sector in a country's GDP. Independent variables: EmProt is an OECD measure of the strength of employment protection institutions and policies which ranges from 1 to 20; Mskill1 is log number of males out of working age male population with low education relative to stock of physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with high education relative to physical capital; Fskill1-Fskill3 are female educational endowments defined analogously; Arable is log arable land area relative to physical capital; P10 is log price of industry 10 output (Agriculture) relative to industry 60 is Business Services; TFP10 is a log superlative index number measure of TFP in industry 10 relative to TFP in industry 30, and so on for the other industries. To abstract from short-run fluctuations all TFP measures are smoothed with a Hodrick-Prescott filter.

The long-run coefficients on relative prices, technology, and factor endowments evaluated at mean levels of employment protection can be obtained from Table 7 by dividing the estimated coefficients by $(1-\delta_0)$ (0.41 in Manufacturing). These are the long-run coefficients reported in Table 8. In general, they correspond closely to the parameter estimates from the static specification in Tables 5 and 6. For example, the estimated long-run coefficient of 0.0414 on male low education corresponds to a value of 0.0355 in Table 5 and 0.0453 in Table 6. The pattern of estimated coefficients on the relative price and TFP terms again conforms to the predictions of theory: the estimated coefficients on own-industry relative prices and TFP in Tables 7 and 8 are all positive (where the estimated manufacturing coefficient is minus the sum of the estimated coefficients on the other industry terms).

GDP share _{ct}	(1) Agriculture	(2) Manufact.	(3) Other	(4) Other	(5) Business
			Production	Services	Services
Obs	200	200	200	200	200
Years	1976-94	1976-94	1976-94	1976-94	1976-94
Mskill1 _{ct} $(\gamma_1/(1-\delta_0))$	0.0414	-0.0881	0.0638	0.0155	0.0075
Mskill2 _{ct} $(\gamma_2/(1-\delta_0))$	0.0159	0.0385	-0.0609	0.1242	-0.1547
Mskill3 _{ct} $(\gamma_3/(1-\delta_0))$	0.0033	-0.0233	0.0382	0.0142	0.0041
Fkill1 _{ct} $(\gamma_4/(1-\delta_0))$	-0.0425	0.0681	-0.0710	0.0114	-0.0077
Fkill2 _{ct} $(\gamma_5/(1-\delta_0))$	-0.0083	-0.0399	0.0409	-0.0231	0.1051
Fkill3 _{ct} $(\gamma_6/(1-\delta_0))$	0.0051	-0.0088	-0.0620	0.0075	0.0130
Arable _{ct} $(\gamma_7/(1-\delta_0))$	0.0044	0.1401	0.0093	-0.1558	-0.0011
P10 _{ct} $(\alpha_1/(1-\delta_0))$	0.0343	-0.0043	-0.0531	0.0414	0.0083
P40 _{ct} $(\alpha_2/(1-\delta_0))$	-0.0073	-0.0627	0.1206	-0.0366	-0.0217
P50 _{ct} $(\alpha_3/(1-\delta_0))$	-0.0003	0.0482	-0.0289	0.0464	-0.0849
P60 _{ct} $(\alpha_4/(1-\delta_0))$	-0.0043	-0.0780	-0.0109	-0.0618	0.1489
TFP10 _{ct} ($\beta_1/(1-\delta_0)$)	0.0185	-0.0581	0.0029	0.0166	0.0053
TFP40 _{ct} ($\beta_2/(1-\delta_0)$)	0.0038	-0.0317	0.1182	-0.0269	-0.0452
TFP50 _{ct} ($\beta_3/(1-\delta_0)$)	0.0087	-0.0767	-0.0715	0.0340	-0.0204
TFP60 _{ct} ($\beta_4/(1-\delta_0)$)	-0.0006	-0.0868	0.0368	-0.0017	0.0626

 Table 8 : Implied Long-run Estimated Coefficients at Mean Values of

 Employment Protection

Notes: long-run estimated coefficients evaluated at the mean value of employment protection across countries from the dynamic within groups specification in Table 7. Since the employment protection variable in the interaction term is normalised by its mean across countries, these long-run coefficients are obtained from the parameter estimates reported in Table 7 by dividing by 1 minus the estimated coefficient on the lagged dependent variable $(1-\delta_0)$. Mskill1 is log number of males out of working age male population with low education relative to stock of physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with high education relative to physical capital; Fskill1-Fskill3 are female educational endowments defined analogously; Arable is log arable land area relative to physical capital; P10 is log price of industry 10 output (Agriculture) relative to industry 30 output (Manufacturing), and so on for the other industries; industry 40 is Other Production; industry 50 is Other Services; industry 60 is Business Services; TFP10 is a log superlative index number measure of TFP in industry 10 relative to TFP in industry 30, and so on for the other industries. To abstract from short-run fluctuations all TFP measures are smoothed with a Hodrick-Prescott filter.

Analyzing the pattern of long-run coefficients across industries, we see that moving a man from low to medium education reduces the share of Agriculture and Other Production in GDP but increases the share of Manufacturing and Other Services. In contrast, the general equilibrium effect of moving a woman from low to medium education is to reduce specialization in Manufacturing but increase specialization in Business Services. Similarly, moving a man from medium to high education increases specialization in Business Services, while moving a woman from medium to high education has exactly the opposite effect. Thus, production structure responds very differently to the educational attainment of men and women, and these results emphasize the importance of distinguishing between male and female labour market outcomes. With many factors of production plus an uneven number of industries and factors of production, there is no one appropriate measure of factor intensity. However, taking all industries together, the pattern of estimated coefficients is broadly consistent with information on factor intensity. For example, low education men account for a relatively large share of total employment in Other Production, while women account for a very small share of employment in this industry.

Table 9 evaluates the contribution of each of the explanatory variables to observed changes in shares of GDP during the sample period. For simplicity, we focus on the static specification excluding the lagged dependent variable, and take the instrumental variables estimates reported in Table 6 (results using the within groups estimates from Table 5 are extremely similar). Taking differences between the beginning and end of the sample period in equation (6), the change in GDP shares predicted by the model can be compared with the change in actual GDP shares, and the predicted changes can be decomposed into the contributions of factor endowments, relative prices, TFP, and the year effects. Table 9 reports the results of such an analysis. In the interests of comparability, the same time-period of 1976-93 is used for Japan, United Kingdom, and United States, while data availability constrains the time-period for West Germany to 1985-93. Since equation (6) is log-linear, the contributions of the individual explanatory variables sum to change in predicted GDP shares.

Looking across industries, the predicted change in GDP shares typically lies close to the actual change, providing evidence that the model is relatively successful in explaining changes in specialization over time. For example, the average absolute prediction error across countries in Manufacturing in Table 9 is 1.05%, compared to an average initial share of Manufacturing in GDP in the 4 countries of 28.74%. The model is least successful in Other Production, which is consistent with the existence of unobserved changes in known mineral resources that are important for this sector. The year effects play a substantial role in all sectors except Business Services, suggesting the importance of common changes in relative prices, TFP, and factor endowments across the 14 OECD countries during the sample period. This is consistent with the idea of de-industrialization as a secular trend or shared experience across OECD countries.

	(1) Agriculture	(2) Manufact.	(3) Other Production	(4) Business Services	(5) Other Services
West Germany (1985-93)					
Actual Change in GDP Share	-0.71	-5.46	-0.78	0.38	6.34
Predicted Change in GDP Share	-0.92	-4.54	-2.26	2.38	5.25
Education (Male + Female)	1.41	-1.10	-1.12	-0.39	3.25
Capital	0.14	-0.39	-0.42	0.71	0.61
Arable Land	0.04	-0.17	-0.13	-0.01	0.35
TFP	0.38	1.51	0.31	0.12	-1.01
Prices	-1.33	0.00	-1.32	0.68	1.12
Year Effects	-1.56	-4.39	0.43	1.27	0.94
Japan (1976-93)					
Actual Change in GDP Share	-3.14	-4.93	1.81	3.43	5.43
Predicted Change in GDP Share	-3.45	-5.47	2.50	3.88	4.57
Education (Male + Female)	1.89	0.38	-3.56	0.95	3.70
Capital	0.71	-1.90	-2.07	3.53	3.02
Arable Land	0.15	-0.67	-0.52	-0.03	1.36
TFP	-2.01	2.55	-1.54	-1.01	0.62
Prices	-1.01	-0.51	6.68	-1.15	-1.46
Year Effects	-3.17	-5.33	3.51	1.58	-2.67
United Kingdom (1976-93)					
Actual Change in GDP Share	-0.92	-7.65	-2.14	8.48	3.97
Predicted Change in GDP Share	-1.30	-9.16	0.09	8.88	2.22
Education (Male + Female)	1.98	1.52	-4.43	1.91	3.56
Capital	0.29	-0.78	-0.85	1.45	1.24
Arable Land	0.20	-0.92	-0.72	-0.04	1.87
TFP	0.25	-4.70	3.87	-1.73	-1.43
Prices	-0.86	1.04	-1.29	5.70	-0.36
Year Effects	-3.17	-5.33	3.51	1.58	-2.67
United States (1976-93)					
Actual Change in GDP Share	-1.39	-5.48	-2.07	8.64	1.22
Predicted Change in GDP Share	-1.89	-6.69	-1.27	8.12	1.98
Education (Male + Female)	1.97	0.89	-2.92	0.32	3.08
Capital	0.32	-0.85	-0.93	1.58	1.35
Arable Land	0.04	-0.18	-0.14	-0.01	0.36
TFP	0.86	-0.03	-1.37	0.58	-0.98
Prices	-1.91	-1.20	0.56	4.06	0.83
Year Effects	-3.17	-5.33	3.51	1.58	-2.67

Table 9: Contribution of Explanatory Variables to Changes in Shares of GDP, Second-stage IV Parameter Estimates from Table 6

Notes: equation (6) is used to decompose the change in shares of GDP into the contributions of changes in the individual explanatory variables between the beginning and end of the sample period. Parameter estimates are taken from the second-stage IV results in Table 6. Education contribution calculated as follows. Multiply the change in the log of each education endowment by its estimated coefficient to yield the predicted effect of changes in that education endowment. Sum across education endowments to give the estimated contribution of changes in education to changes in shares of GDP. Contributions of other variables are calculated analogously. Since equation (6) is log linear, summing the contributions over time for individual country-industries. Changes in shares of GDP and their components are reported as percentages. Thus, in Germany, Agriculture's share of GDP was predicted to fall by 0.92%. The change in relative prices implied a fall in Agriculture's share of GDP of 1.33%. The year effects capture the effect of any change in relative prices, technology, and factor endowments that is common across countries. The time-period over which averages are taken is 1976-93, except for West Germany where the time-period is 1985-93. This variation in time-period explains the difference in the year effects' contribution between West Germany and the other countries.

However, we noted earlier that the timing and magnitude of de-industrialization varies substantially. The decline in manufacturing's share of GDP occurred earlier and was more extensive in the United Kingdom and United States than in Germany and Japan. Table 9 suggests that this is largely explained by differences in rates of TFP growth across the 4 countries. While the net contribution of TFP growth in all 5 sectors to changes in Manufacturing's share of GDP was negative in the United Kingdom and United States (-4.70% and -0.03% respectively), West Germany and Japan experienced positive contributions (1.51% and 2.55% respectively). The especially rapid decline in Agriculture's share of GDP in Japan (3.14%) is largely explained by rates of TFP growth (a contribution of 2.01%), although the evolution of relative prices across sectors also plays an important role (a contribution of 1.01%).

Rising educational attainment made a large negative contribution to the change in Other Production's GDP share and a large positive contribution to the change in Other Services' GDP share. Thus, other things equal, countries with more rapid increases in educational attainment experienced larger rises in the share of Other Services in GDP, and this is consistent with the idea that many service sectors are relatively skill intensive. Table 9 suggests that the more rapid increase in Business Services' GDP share in the United Kingdom and United States relative to West Germany and Japan was largely due to country-specific changes in relative prices (which, for example, made a contribution of 5.70% in the United Kingdom and –1.15% in Japan). Physical capital also made a positive contribution to the expansion of Business Services and Other Services, and this is consistent with the extremely high values of Buildings and Structures capital (in particular, real estate) in these sectors.

6. Conclusions

A key feature of economic growth in industrialized countries since the early 1970s has been the secular decline in manufacturing's share of GDP and the secular rise in the share of service sectors. Although these changes are common to all OECD countries, their magnitude and timing varies substantially. This paper examines the role played by relative prices, technology, factor endowments, and labour market institutions in the process of 'de-industrialization'.

We find a statistically significant and quantitatively important effect of levels of educational attainment on patterns of production. The effect of a given level of educational attainment varies substantially between men and women. While moving a man from low to medium education reduces the share of Agriculture and Other Production in GDP and increases the share of Manufacturing and Other Services, moving a woman from low to medium education reduces specialization in Manufacturing and increases specialization in Business Services. These

33

findings are consistent with the large labour market literature on male-female differentials and suggest that men and women of the same educational level have or are perceived to have different vectors of other characteristics or dimensions of skills.

The effects of relative prices and technical efficiency on patterns of production confirm to the predictions of theory. We find positive and statistically significant impact of own-industry prices and technical efficiency on the share of a sector in GDP. In a specification including a country-industry effect and industry-time dummies, we reject the null hypothesis that the coefficients on relative prices are equal to zero, suggesting an important role for country-specific changes in relative prices. The results are robust to instrumenting domestic relative prices using prices at the border, exchange rates, and average tariff levels. We provide evidence that the instruments are highly statistically significant in the first-stage regression and that the identifying assumptions underlying the instrumental variables estimation are satisfied.

There is evidence of partial adjustment towards long-run patterns of specialization, and the speed of adjustment is shown to vary systematically across countries with their extent of provision for employment protection. This confirms the importance of labour market policies and institutions in facilitating the reallocation of resources from declining to expanding sectors. Taken together, our results emphasize the role of labour market characteristics, policies, and institutions in shaping product market outcomes.

Appendix A: Imperfect Competition and External Economies of Scale

Although the neoclassical model is typically analysed under the assumption of perfect competition, it is relatively straightforward to introduce imperfect competition following Helpman (1984). Suppose that the representative consumer's utility function is weakly separable in the output of sectors j=1,...,n, while, within each sector j, a variety of differentiated products are consumed g=1,...,G,

$$U = U[u_1(\cdot), u_2(\cdot), \cdots, u_n(\cdot)]$$
(A1)

$$u_{j}(y_{j1}, \cdots, y_{jG}) = \left(\sum_{g=1}^{G} y_{jg}^{\beta_{j}}\right)^{1/\beta_{j}}, \qquad 0 < \beta_{j} < 1$$
(A2)

where, for simplicity, we assume that the sub-utility function $u_j(A)$ takes the Dixit-Stiglitz form. Profit maximization yields the following standard result,

$$p_{jg} = p_j = \left(\frac{\sigma_j}{\sigma_j - 1}\right) b_j(q) \tag{A3}$$

where $\sigma_j = 1/(1-\beta_j)$ is the perceived elasticity of demand facing the producer of each variety in sector *j*; $b_j(\cdot)$ is the unit cost function – assumed to be the same for all producers in a given sector; *q* is the vector of factor prices. The revenue function is again defined as the value of profit maximising outputs (r(p,v)=p.y(p,v)), where the value of output of good *j* is simply

$$\sum_{g} p_{jg} \cdot y_{jg} = p_{j} \cdot \sum_{g} y_{jg} \cdot$$

The analysis may also be extended to introduce external economies of scale or localized knowledge spillovers (see Helpman (1984)).³¹ This corresponds to the case where the country-industry technology variable (θ_{gi}) in Section 2 is a function of the scale or density of economic activity in a particular location (as in Ciccone and Hall (1996)).

³¹ See Davis and Weinstein (1998b), (1999), Fujita, Krugman, and Venables (1999), and Midelfart-Knarvik, Overman, and Venables (2000) for work combining factor endowments and economic geography as determinants of production structure.

Appendix B: Data Sources

B.1 Summary of educational attainment data sources

Australia: 1982-1993 & 1979-93, male & female separately, population and labour force.

Data for labour force and population male & female separately from the "Labour Statistics Australia ABS Catalogue No.6101.0".

Actual population & labour force data for the years 1982-1993 & 1979-93 respectively.

Three groups: **(low)** Didn't attend highest level of Secondary School **(middle)** Attended highest level of Secondary School + Trade, Tech or Certificate **(high)** Degree

Belgium: 1986-1995 male & female separately, population only.

Data for the population male & female separately from "Enquete sur les forces de travail" Statistiques Sociales, Ministere des Affaires Economiques 1995 & 1991.

Actual population data for all the years.

Three groups: (low) Enseignement primaire ou non (middle) Enseignement secondaire inferieur + Enseignement superieur non universitaire de type court + Enseignement superieur non universitaire de type long (high) Enseignement universitaire

Canada: 1975-1995 male & female separately, population and labour force.

Data for labour force and population male & female separately from the "Annual Labour Force Averages 1981-1988 & 1989-1994 & 1995 & 1996" & "The Labour Force: 1975, 1976, 1977, 1978 & 1979".

Actual population & labour force data for all the years but with some data replaced by interpolations for 1990-95.

Three groups (pre 1981): (low) 0-8 years (middle) Some high school & no post-secondary + Some postsecondary + Post-secondary certificate or diploma (high) University Degree

Three groups (1981-89): (low) 0-8 years (middle) 9-13 years + Some post-secondary + Post-secondary certificate or diploma (high) University Degree

Three groups (1990-96): (low) 0-8 years (middle) 9-13 years + Graduated from high school + Some post-secondary + Post-secondary certificate or diploma (high) University Degree

Denmark: 1983-1995, male & female separately, population only.

Data for population male & female separately from the Danish Annual Statistics "Statistik Arbog".

Actual population data for the years 1983, 1985, 1987, 1988, 1990, 1991, 1993 & 1995. Missing years of data were filled in by interpolation between 1983 and 1995 and but with some data replaced by interpolations for 1991-95.

Three groups: (low) second level, first stage (middle) second level, second stage (general and vocational) and education at the third level, first stage (university and non-university type) (high) third level, second stage.

France: 1982-1995, male & female separately, labour force only.

Data for labour force male & female separately from the "Enquete sur l'emploi: resultats detaille", les collections de l'insee, institut national de la statistique et des etudes economiques.

Actual labour force data for the years 1982, 1983, 1986-90, 1992, 1993, 1994 & 1997 for men and women separately. Missing years of data were filled in by interpolation between 1982 and 1997.

Three groups (pre 1994): **(low)** Aucun diplome ou certificat d'etudes (cep) seul **(middle)** Brevet d'etudes du premier cycle (bepc) seul + Cap, bep, ou autre diplome de ce niveau + Baccalaureat ou brevet professional, ou autre diplome de ce niveau **(high)** Diplome du 1er cycle universitaire, bts, dut, diplome paramedical ou social + Diplome du 2e or 3e cycle universitiaire, diplome d'une grande ecole ou ecole d'ingenier

Three groups (1994 onwards): (low) Aucun diplome ou CEP (middle) BEPC seul + Cap, bep, ou autre diplome de ce niveau + Baccalaureat, brevet professional ou autre diplome de ce niveau (high) Baccalaureat + 2 ans + Diplome superieur

Finland: 1984-1995 male & female separately, population and labour force.

Data for labour force and population male & female separately from the "Tyovoiman koulutus ja ammatit" Statistics Finland.

Actual population & labour force data for all the years.

Three groups: **(low)** Basic education only **(middle)** Upper secondary education - lower level + Upper secondary education – upper level **(high)** Higher education - lower level + Higher education – upper level

Germany: 1984-1995 male & female separately, population and labour force.

Data for the population and labour force, male & female separately from German Socio-Economic Panel (GSOEP).

Actual population & labour force data for all years but with some data replaced by interpolations for 1991-95.

Three groups: (low) 9 years of schooling + no vocational (middle) 10 years of schooling and no vocational + 9 years of schooling and some vocational + 10 years of schooling and some vocational + 13 years of schooling + College (high) University

Thanks to Damon Clark (CEP, LSE)

Italy: 1977-1995 male & female separately, labour force only.

Data for the population male & female separately from "Indagine sulle forze di lavoro" in the table "Forze di lavoro secondo di sesso, la classe di eta, il titolo di studio e la condizione".

Actual labour force data for all years but with some data replaced by interpolations for 1993-95.

Three groups (pre 1993): **(low)** Senza titolo e licenza elementare **(middle)** Licenza scuola media inferiore + Licenza scuola media superiore **(high)** Laurea

Three groups (1993+): (low) Dottorato di ricerca o Specializazione + Laurea (middle) Diploma Univ. o Laureabreve + Diploma accesso Universita + Qualifica Lic. Non accesso Universita + Licenza media (high) Licenza elementare, nessun titolo

Thanks to Marco Manacorda (Berkeley, USA) and Barbara Petrongolo (Madrid, Spain)

Japan: 1975-1995, male & female separately, labour force only.

Data for labour force male & female separately from the Employment Status Survey published by the Statistics Bureau, Management and Coordination Agency, Government of Japan.

Actual labour force data for the years 1971, 1974, 1979, 1982, 1987, 1992 & 1997 for men and women separately. Missing years of data were filled in by interpolation between 1971 and 1997.

Three groups: (low pre 1980) never attended, elementary school and Junior high school (low post 1980) elementary school and Junior high school (middle) Senior high school and Junior college, technical college (high) College or university, including graduate school.

Thanks to Toshiaki Tachibanaki (KIER, Kyoto University) for supplying us directly with these figures.

Netherlands: 1975-1995 male & female separately, population and labour force.

Data for the population male & female separately from the Population Survey (Volkstelling), the Annual Labour Force Survey (Arbeidskrachtentelling (AKT)) and the Monthly Labour Force Survey (Enquete Beroepsbevolking (EBB)).

Actual population & labour force data for the years 1975, 1977, 1979, 1981, 1983, 1985, 1988-95. Missing years of data were filled in by interpolation between 1975 and 1995.

Three groups: (low) no qualifications (middle) lower + intermediate (high) higher qualifications

Thanks to Jan van Ours (Tilburg, Netherlands) for supplying us directly with these figures.

Norway: 1975-1995 male & female separately, population only.

Data for the population male & female separately from Statistical Yearbook of Norway, Statistics Norway. Actual population data for the years 1975, 1980, 1985, 1986, 1988, 1989 & 1990-95. Missing years of data were filled in by interpolation between 1975 and 1995

Three groups: (low) Unknown or no completed education (middle) Education at the second level, first stage + Education at the second level, second stage (high) Education at the third stage.

Sweden: 1975-1995 male & female separately, population only.

Data for the population male & female separately from Swedish Labour Force Survey.

Actual population data for all the years.

Three groups: (low) Elementary school (< 9 years) + 9-year compulsory school (middle) Upper secondary school, 2 years or shorter + Upper secondary school, 3 years (high) Tertiary (post-secondary) education, 3 years or longer + postgraduate education

Thanks to Gunilla Dahlén (SCB, Sweden) and Ingrid Turtola (SCB, Sweden)

UK: 1975-1995 male & female separately, population and labour force.

Data for the population and labour force with male & female separately from General Household Survey. Actual population & labour force data for all the years.

Three groups (1977-90): (low) no qualifications (middle) voc-high + teaching + nursing + A-level + voc-middle + O-level 5+ + voc-low + O-level & clerical + O-level 1-4 + clerical + voc-other + other (high) University

Three groups (1991-95): **(low)** no qualifications **(middle)** Teaching + other high + nursing + gce a level + gcse & olevel + gcse & olevel + gcse & olevel + comm q, n + cse grd + apprenticeship + scst g6- + foreign + other qual **(high)** higher degree + first degree

Thanks to Glenda Quintini (CEP, LSE) and Steve McIntosh (CEP, LSE)

USA: 1975-1995 male & female separately, population and labour force.

Data for the population and labour force for male & female separately are from the Current Population Survey (CPS).

Actual population & labour force data for all years except 1976 and 1978 (for population data) and some data replaced by interpolations for 1992-95.

Three groups: (low) non high school graduates (middle) high school graduates + college and associated degrees (high) bachelors and higher

Thanks to Lupin Rahman (CEP, LSE) and Randy E. Ilg (Bureau of Labor Statistics, USA)

B2. Production Data and Other Independent Variables

OECD International Sectoral Database (ISDB): data on current price value-added, real value-added (1990 US dollars), real physical capital stock (1990 US dollars), employment, and hours worked for the one-digit industries listed in Table 1B in the main text for the years 1976-94. Data on current price GDP and aggregate real physical stock (1990 US dollars) for 1976-94.

United Nations FAO: data on arable land area (thousands of hectares) for 1970-94.

OECD Bilateral Trade Database: data on bilateral imports between the 14 OECD countries for 1970-94 used to construct prices at the border.

OECD Structural Analysis Industrial Database (STAN): data on nominal exchange rates for converting prices at the border to national currencies.

IMF International Financial Statistics, IMF Government Finance Statistics, and Annual Reports of the European Commission: data on the ratio of tariff revenues to the value of imports. See Djankov *et al.* (1999) for further information concerning these data.

OECD Jobs Study: Evidence and Explanations: index of the strength of employment protection institutions and policies. See Nickell (1997) for further information concerning these data.

Code	ISIC	Industry
10	10	Agriculture, Hunting, Forestry, and Fishing
	11	Agriculture and Hunting
	11.1	Agriculture and Livestock Production
	11.2	Agricultural Services
	11.3	Hunting, Trapping, and Game Propagation
	12	Forestry and Logging
	12.1	Forestry
	12.2	Logging
	13	Fishing
	1301	Ocean and Coastal Fishing
	1302	Fishing Not Elsewhere Classified
30	30	Manufacturing
	31	Food, Beverages, and Tobacco
	32	Textile, Wearing Apparel, and Leather Industries
	33	Wood and Wood Products, Including Furniture
	34	Paper and Paper Products; Printing and Publishing
	35	Chemicals and Chemical Products; Petroleum, Coal, Rubber, and Plastic
	36	Non-metallic Mineral Products, except Petroleum and Coal
	37	Basic Metal Industries
	38	Fabricated Metal Products, Machinery and Equipment
	39	Other Manufacturing Industries
40		Other Production
	20	Mining and Quarrying
	21	Coal Mining
	22	Crude Petroleum and Natural Gas Production
	23	Metal Ore Mining
	29	Other Mining
	40	Electricity, Gas, and Water
	41	Electricity, Gas, and Steam
	42	Water Works and Suppy
	50	Construction
		Construction of Dwellings
		Construction of Non-residential Buildings
		Civil Engineering Works
		Demolition of Buildings
602	80	Business Services
	81	Financial Institutions
	8101	Monetary Institutions
	8102	Other Financial Institutions
	8103	Financial Services
	82	Insurance
	83	Real Estate and Business Services
	831	Real Estate
	832	Business Services Except Machinery and Equipment Rentals and Leasing
	833	Machinery and Equipment Rental and Leasing

Table B1: Breakdown of the Disaggregated Sectors Included in each One-Digit Industry (International Standard Industrial Classification (ISIC))

Code	ISIC	Industry
601		Other Services
	60	Wholesale and Retail Trade, Restaurants and Hotels
	61	Wholesale Trade
	62	Retail Trade
	63	Restaurants and Hotels
	631	Restaurants, Cafes, and Other Eating and Drinking Places
	632	Hotels, Rooming Houses, Camps, and Other Lodging Places
	70	Transport, Storage, and Communication
	71	Transport and Storage
	711	Land Transport
	712	Water Transport
	713	Air Transport
	719	Services Allied to Transport
	72	Communication
	90	Community, Social, and Personal Services
	91	Public Administration and Defence
	92	Sanitary and Similar Services
	93	Social and Related Community Services
	931	Education Services
	932	Research and Scientific Institutes
	933	Medical, Dental, Other Health, and Veterinary Services
	934	Welfare Institutions
	935	Business, Professional, and Labour Associations
	939	Other Social and Related Community Services
	94	Recreational and Cultural Services
	941	Motion Picture and Other Entertainment Services
	942	Libraries, Museums, Botanical Gardens, and Other Cultural Services nes
	949	Amusement and Recreational Services nes
	95	Personal and Household Services
	951	Repair Services nes
	96	International and Other Extra-territorial Bodies
100		Producers of Government Services and Other Producers
		Producers of Government Services
		Other Producers
1000		Total Including All Taxes

Table B1 (cont): Breakdown of the Disaggregated Sectors Included in each One-Digit Industry (International Standard Industrial Classification (ISIC))

Industry	Year	Percentage
Agriculture		
Male	1975	84.4
	1985	83.0
	1995	79.4
Female	1975	15.6
	1985	17.0
	1995	20.6
Manufacturing		
Male	1975	69.6
	1985	71.5
	1995	71.0
Female	1975	30.4
	1985	28.5
	1995	29.0
Other Production		
Male	1975	86.0
	1985	85.5
	1995	80.8
Female	1975	14.0
	1985	14.5
	1995	19.2
Business Services		
Male	1975	52.5
	1985	46.6
	1995	42.2
Female	1975	47.5
	1985	53.4
	1995	57.8
Other Services		
Male	1975	53.5
	1985	50.4
	1995	50.0
Female	1975	46.5
	1985	49.6
	1995	50.0

Table B2: Female & Male Employment as a Percentage of Total Employment, United Kingdom, 1975, 1985 & 1995

Notes: for each year and each industry, the male employment percentage is number of male employees divided by total number of employees (male plus female) expressed as a percentage. The female employment percentage is defined analogously. Source: United Kingdom New Earnings Survey.

Industry	Year	Low	Medium	High
Agriculture				
Male	1975	3.03 (2.56)	92.36 (77.94)	4.61 <i>(3.89)</i>
	1985	3.12 (2.59)	92.57 (76.80)	4.30 (3.57)
	1995	1.89 <i>(1.50)</i>	94.25 (74.80)	3.86 (3.07)
Female	1975	5.12 (0.80)	89.42 (13.96)	5.46 (0.85)
remaie	1975			
		3.38 (0.58)	91.89 (15.66)	4.73 (0.81)
	1995	3.63 (0.75)	92.74 <i>(19.14)</i>	3.63 (0.75)
Manufacturing				
Male	1975	6.07 <i>(4.23)</i>	82.71 (57.60)	11.22 (7.82)
	1985	5.33 (3.81)	80.44 (57.50)	14.23 (10.17)
	1995	3.81 (2.70)	77.51 (55.05)	18.68 (13.27)
Female	1975	6.74 (2.05)	90.18 (27.37)	3.08 (0.93)
1 ennaie	1985	5.31 (1.51)	90.59 (25.84)	4.11 (1.17)
	1995	3.45 (1.00)	89.90 (26.05)	6.65 (1.93)
	1995	5.45 (1.00)	89.90 (20.09)	0.05 (1.99)
Other Production				
Male	1975	12.63 <i>(10.86)</i>	75.52 <i>(64.95)</i>	11.85 <i>(10.20)</i>
	1985	12.43 (10.63)	73.82 <i>(63.13)</i>	13.74 <i>(11.75)</i>
	1995	8.56 <i>(6.91)</i>	68.12 <i>(55.04)</i>	23.32 (18.84)
Female	1975	9.25 (1.29)	86.29 (12.07)	4.46 (0.62)
	1985	8.38 (1.21)	84.74 (12.27)	6.88 (1.00)
	1995	4.15 (0.80)	84.86 (16.30)	10.99 (2.11)
	1775	1.15 (0.00)	01.00 (70.90)	10.99 (2.17)
Business Services				
Male	1975	11.31 <i>(5.94)</i>	70.39 <i>(36.95)</i>	18.31 <i>(9.61)</i>
	1985	10.70 <i>(4.99)</i>	69.88 <i>(32.58)</i>	19.43 <i>(9.06)</i>
	1995	11.05 (4.67)	67.74 <i>(28.60)</i>	21.20 (8.95)
Female	1975	16.05 (7.62)	70.50 <i>(33.49)</i>	13.45 (6.39)
	1985	15.38 (8.21)	70.13 (37.43)	14.49 (7.73)
	1995	12.25 (7.08)	72.80 (42.06)	14.95 (8.64)
Other Services				
Male	1975	5.21 (2.79)	73.03 (39.08)	21.75 (11.64)
1.1uic	1985	3.49 (1.76)	71.28 (35.91)	25.23 (12.71)
	1995	2.87 (1.44)	67.95 <i>(34.01)</i>	29.18 (14.60)
Eo	1075	4 24 (1 07)	00.67 (42.15)	5.00 (2.27)
Female	1975	4.24 (1.97)	90.67 (42.15)	5.09 (2.37)
	1985	2.99 (1.48)	90.90 (45.10)	6.11 (3.03)
	1995	2.53 (1.26)	86.18 <i>(43.05)</i>	11.29 (5.64)

Table B3: Percentages of Female & Male Employment by Level of Educational Attainment, United Kingdom, 1975, 1985 & 1995

Notes: the value 3.03 in the top left-hand cell is the number of men with low education in Agriculture in 1975 expressed as a percentage of all male employees. The values for women and for the other cells are defined analogously. The figures in parenthesis are expressed as a percentage of all employees. The value 2.56 in the top left-hand cell is thus the number of men with low education in Agriculture in 1975 expressed as a percentage of all employees (male plus female). The values for women and for the other cells are defined analogously. Source: United Kingdom New Earnings Survey.

Appendix C: Panel Data Unit Root Tests

Levi	n-Lin	Madd	ala-Wu
Variable	Test Statistic	Variable	Test Statistic
Mskill1 _{ct}	-2.071	Mskill1 _{ct}	73.011
Mskill2 _{ct}	-5.485	Mskill2 _{ct}	15.299
Mskill3 _{ct}	-5.948	Mskill3 _{ct}	90.047
Fskill1 _{ct}	-1.776	Fskill1 _{ct}	17.621
Fskill2 _{ct}	-8.166	Fskill2 _{ct}	20.411
Fskill3 _{ct}	-7.460	Fskill3 _{ct}	101.824
Arable _{ct}	-2.592	Arable _{ct}	50.352
P10 _{ct}	-9.011	P10 _{ct}	13.301
$P40_{ct}$	-6.339	P40 _{ct}	32.670
P50 _{ct}	-5.094	P50 _{ct}	39.438
P60 _{ct}	-5.214	P60 _{ct}	8.813
TFP10 _{ct}	-5.364	TFP10 _{ct}	115.300
$\mathrm{TFP40}_{\mathrm{ct}}$	-4.117	$\mathrm{TFP40}_{ct}$	17.145
TFP50 _{ct}	-6.626	$\mathrm{TFP50}_{\mathrm{ct}}$	23.696
TFP60 _{ct}	-1.854	TFP60 _{ct}	94.296
Gdpsh10 _{ct}	-5.974	Gdpsh10 _{ct}	19.432
Gdpsh30 _{ct}	-2.371	Gdpsh30 _{ct}	22.345
Gdpsh40 _{ct}	-1.734	Gdpsh40 _{ct}	53.389
Gdpsh50 _{ct}	-1.762	Gdpsh50 _{ct}	44.240
Gdpsh60 _{ct}	-0.607	Gdpsh60 _{ct}	36.748

Table C1: Unit Root Tests (Levin and Lin (1992) and Maddala and Wu (1999))

Notes: Mskill1 is log number of males out of working age male population with low education relative to stock of physical capital; Mskill2 is log number of males with medium education relative to physical capital; Mskill3 is log number of males with high education relative to physical capital; Fskill1-Fskill3 are female educational endowments defined analogously; Arable is log arable land area relative to physical capital; P10 is log price of industry 10 output (Agriculture) relative to industry 30 output (Manufacturing), and so on for the other industries; industry 40 is Other Production; industry 50 is Other Services; industry 60 is Business Services; TFP10 is a log superlative index number measure of TFP in industry 10 relative to TFP in industry 30, and so on for the other industries. Gdpsh10 is share of industry 10 in country GDP, and so on for the other industries. Δ denotes the difference operator. Equation estimated for Levin and Lin (1992) panel data unit root test is $y_{\alpha} = \rho y_{\alpha,1} + \eta_{\epsilon} + \varepsilon_{\alpha}$. Null hypothesis, $H_0: \rho = 1$. Critical values are taken from Levin and Lin op cit., Table 5: 1% critical value = -6.72; 5% critical value = -6.28; 10% critical value = -6.04. Equation estimated for Maddala and Wu (1999) panel data unit root test is: $\Delta y_d = \rho_e y_{d,1} + \rho_e y_{d,2}$ $\alpha_1 \Delta y_{\alpha,1} + \Delta \alpha_2 \Delta y_{\alpha,2} + \eta_c + \varepsilon_a$ (an Augmented Dickey-Fuller specification). Equation estimated separately for each country. Null hypothesis is $H_0: \rho_c = 0$ for c = 1, 2, ..., C, against the alternative $H_1: \rho_c \neq 0$. Denote the MacKinnon approximate *p-value* for the parameter ρ_e in the regression for an individual country by π_e (e=1,2...,C). Following Fisher (1932), Maddala and Wu op cit. note that $\lambda = -2 \mathcal{J}_c \ln(\pi_c)$ is distributed $\chi^2(2C)$. Chi-squared (28) critical values are 1% : 48.2782; 5% : 41.3372; 10% : 37.9159.

References

Ashton, D, Davies, B, Felstead, A, and Green, F (1998) *SKOPE Monograph No. 1*, Economic and Social Research Council, University of Warwick, and University of Oxford, United Kingdom.

Barro, R and Lee, J (1993) 'International Comparisons of Educational Attainment', *Journal of Monetary Economics*, 32(3), 363-94.

Barro, R and Lee, J (2000) 'International Data on Educational Attainment: Updates and Implications', *CID Working Paper*, 42.

Baxter, M and King, R (1999) 'Measuring Business Cycles: Approximate Band-Pass Filters for Economic Time Series', *Review of Economics and Statistics*, 81(4), 573-93.

Bayard, K, Hellerstein, J, Neumark, D, and Troske, K (1999) 'New Evidence on Sex Segregation and Sex Differences in Wages from Matched Employee-Employer Data', *NBER Working Paper*, 7003.

Berman, E, Bound, J, and Machin, S (1998) 'Implications of Skill-biased Technological Change: International Evidence', *Quarterly Journal of Economics*, November, 1245-79.

Bernard, A and Jones, C (1996a) 'Productivity across Industries and Countries: Time Series Theory and Evidence', Review of Economics and Statistics, 78(1), 135-46.

Bernard, A and Jones, C (1996b) 'Comparing Apples to Oranges: Productivity Convergence and Measurement Across Industries and Countries', *American Economic Review*, 1,216-38.

Bernstein, J and Weinstein, D (1998) 'Do Endowments Predict the Location of Production? Evidence from National and International Data', *NBER Working Paper*, 6815

Blinder, A (1973) 'Wage Discrimination: Reduced Form and Structural Estimates', *Journal of Human Resources*, 8(4), 436-55.

Blundell, R and MaCurdy, T (2001) 'Labor Supply: A Review of Alternative Approaches', Chapter 27 in (eds) Ashenfelter, O and Card, D, *Handbook of Labor Economics*, Volume 3, Elsevier Science : Netherlands.

Bowen, H, Leamer, E, and Sveikauskas, L (1987) 'Multicountry, Multifactor Tests of the Factor Abundance Theory', *American Economic Review*, 77, 791-809

Broadberry, S (1997) The Productivity Race: British Manufacturing in International Perspective, 1850-1990, Cambridge University Press : Cambridge.

Caves, D, Christensen, L, and Diewert, E (1982a) 'The Economic Theory of Index Numbers and the Measurement of Input, Output, and Productivity', *Econometrica*, 50(6), 1393-1414

Caves, D, Christensen, L, and Diewert, E (1982b) 'Multilateral Comparisons of Output, Input, and Productivity Using Superlative Index Numbers', *Economic Journal*, 92, 73-86

Christensen, L, Jorgenson, D, and Lau, L (1973) 'Transcendental Logarithmic Production Frontiers', *Review of Economics and Statistics*, 55(1), 28-45.

Ciccone, A and Hall, R (1996) 'Productivity and the Density of Economic Activity', *American Economic Review*, 86(1), pages 54-70.

Cohen, D and Soto, M (2001) 'Growth and Human Capital: Good Data, Good Results', CEPR Discussion Paper, 3025.

Crafts, N (1996) 'De-industrialization and Economic Growth', Economic Journal, 106, 172-83.

Davis, D, Weinstein, D, Bradford, S, and Shimpo, K (1997) 'Using International and Japanese Regional Data to Determine When the Factor Abundance Theory of Trade Works', *American Economic Review*, 87(3), 421-46.

Davis, D and Weinstein, D (1998a) 'An Account of Global Factor Trade', NBER Working Paper, 6785, November

Davis, D and Weinstein, D (1998b) 'Market Access, Economic Geography, and Comparative Advantage: An Empirical Assessment', *NBER Working Paper*, 6787.

Davis, D and Weinstein, D (1999) 'Economic Geography and Regional Production Structure: An Empirical Investigation', *European Economic Review*, 43, 379-407.

de la Fuente, A and Domenech, R (2000) 'Human Capital in Growth Regressions: How Much Difference Does Data Quality Make?', Universidad de Valencia, mimeo

Desai, T, Gregg, P, Steer, J, and Wadsworth, J (1999) 'Gender and the Labour Market', in (eds) Gregg, P, and Wadsworth, J, *The State of Working Britain*, Manchester : Manchester University Press.

Dixit, A and Norman, V (1980) The Theory of International Trade, Cambridge: Cambridge University Press.

Djankov, S, Evenett, S, and Yeung, B (1999) 'Have Changes in Technology Increased OECD Trade?', The World Bank, mimeograph.

Feenstra, R and Hanson, G (1999) 'The Impact of Outsourcing and High-Technology Capital on Wages: Estimates for the United States, 1979-1990', *Quarterly Journal of Economics*, August, 907-40.

Fisher, R (1932) Statistical Methods for Research Workers, Oliver and Boyd : Edinburgh.

Fujita, M, Krugman, P, and Venables, A (1999) *The Spatial Economy: Cities, Regions, and International Trade*, MIT Press.

Gabaix, X (1997) 'The Factor Content of Trade: A Rejection of the Heckscher-Ohlin-Leontief Hypothesis', Harvard University, mimeo.

Gandal, N, Hanson, G, and Slaughter, M (1999) 'Rybczynski Effects and Adjustment to Immigration in Israel', Paper presented at CEPR ERWIT conference, mimeograph. Green, F and James, D (2001) 'Do Male Bosses Underestimate Their Female Subordinates' Skills? A Comparison of Employees' and Line Managers' Perceptions of Job Skills', *Department of Economics Discussion Paper*, 01/07, University of Kent, United Kingdom.

Griffith, R, Redding, S, and Van Reenen, J (2000) 'Mapping the Two Faces of R&D: Productivity Growth in a Panel of OECD Industries', *CEPR Discussion Paper*, 2457.

Hall, R (1990) 'Invariance Properties of Solow's Productivity Residual', in (eds) Diamond, P, *Growth/Productivity/Unemployment: Essays to Celebrate Bob Solow's Birthday*, MIT Press : Cambridge MA, 71-112.

Hanson, H and Slaughter, M (1999) 'The Rybczynski Theorem, Factor-Price Equalisation, and Immigration: Evidence from US States', *NBER Working Paper*, 7074

Harrigan, J (1995) 'Factor Endowments and the International Location of Production: Econometric Evidence for the OECD, 1970-85', *Journal of International Economics*, 39, 123-41

Harrigan, J (1997) 'Technology, Factor Supplies, and International Specialisation: Estimating the Neoclassical Model', *American Economic Review*, 87(4), 475-94.

Harrigan, J (1999) 'Estimation of Cross-country Differences in Industry Production Functions', *Journal of International Economics*, 47(2), 267-93.

Harrigan, J and Zakrajsek, E (2000) 'Factor Supplies and Specialisation in the World Economy', NBER Working Paper, 7848.

Hausman, J (1978) 'Specification Tests in Econometrics', Econometrica, 46, 1251-71.

Helpman, E (1984) 'Increasing Returns, Imperfect Markets, and Trade Theory', Chapter 7, in (eds) Jones, R, and Kenen, P, *Handbook of International Economics*, vol I, 325-65.

Helpman, E (1999) 'The Structure of Foreign Trade', Journal of Economic Perspectives, 13(2), 121-44.

Hodrick, R and Prescott, E (1997) 'Postwar U.S. Business Cycles: An Empirical Investigation', *Journal of Money, Credit, and Banking*, 29(1), 1-16.

Hopenhayn, H and Rogerson, R (1993) 'Job Turnover and Policy Evaluation: A General Equilibrium Analysis', Journal of Political Economy, 101(5), 915-38.

Im, K, Pesaran, M, and Shin, Y (1997) 'Testing for Unit Roots in Heterogeneous Panels', University of Cambridge, mimeo.

Islam, N (1995) 'Growth Empirics: a Panel Data Approach', *Quarterly Journal of Economics*, 110, 1127-70.

Jaffe, A, Trajtenberg, M, and Henderson, R (1993) 'Geographic Localisation of Knowledge Spillovers as Evidenced by Patent Citations', *Quarterly Journal of Economics*, August, 577-98.

Katz, L and Murphy, K (1992) 'Changes in Relative Wages, 1963-87: Supply and Demand Factors', *Quarterly Journal of Economics*, February, 35-78.

Killingsworth, M and Heckman, J (1986) 'Female Labor Supply: A Survey', Chapter 2 in (eds) Ashenfelter, O and Layard, R, *Handbook of Labor Economics*, Volume 1, 103-????, Elsevier Science : Netherlands.

Kitson, M and Michie, J (1996) 'Britain's Industrial Performance Since 1960: Under-investment and Relative Decline, *Economic Journal*, 196-212.

Kohli, U (1991) Technology, Duality and Foreign Trade, Ann Arbor: University of Michigan Press

Krueger, A and Lindahl, M (2000) 'Education for Growth: Why and For Whom?', NBER Working Paper, 7591

Krugman, P (1995) 'Growing World Trade: Causes and Consequences', *Brookings Papers on Economic Activity*, 1, Cambridge Massachusetts, 327-62.

Lawrence, R and Slaughter, M (1993) 'International Trade and American Wages in the 1980s: Giant Sucking Sound or Small Hiccup ?', *Brookings Papers on Economic Activity*, 2, 161-210

Lazear, E (1990) 'Job Security Provisions and Employment', *Quarterly Journal of Economics*, August, 699-726.

Learner, E (1984) Sources of Comparative Advantage: Theories and Evidence, MIT Press.

Leamer, E (1998) 'In Search of Stolper-Samuelson Linkages Between International Trade and Lower Wages', Chapter 4 in (ed) Collins, S, *Imports, Exports, and the American Worker*, Brookings Institution Press

Leamer, E and Levinsohn, J (1995) 'International Trade Theory: The Evidence', *Handbook of International Economics*, vol. 3, 1339-94.

Levin, A and Lin, C (1992) 'Unit Root Tests in Panel Data: Asymptotic and Finitesample Properties', University of California, San Diego, Discussion Paper, 92-23.

Machin, S and Van Reenen, J (1998) 'Technology and Changes in Skill Structure: Evidence from Seven OECD Countries', *Quarterly Journal of Economics*, November, 1215-44.

Maddala, G and Wu, S (1999) 'A Comparative Study of Unit Root Tests with Panel Data and a New Simple Test', *Oxford Bulletin of Economics and Statistics*, Special Issue, 631-52.

Maskus, K, Sveikauskas, C, and Webster, A (1994) 'The Composition of the Human Capital Stock and its Relation to International Trade: Evidence from the United States and Britain', *Weltwirtschaftliches Archiv*, 130, 50-76

Maskus, K and Webster, A (1999) 'Estimating the HOV Model with Technology Differences Using Disaggregated Labor Skills for the United States and United Kingdom', *Review of International Economics*, 7(1), 8-19 Matthews, R, Feinstein, C, and Odling-Smee, J (1982) *British Economic Growth, 1856-1973*, Clarenden Press : Oxford.

Midelfart-Knarvik, K, Overman, H, and Venables, A (2000) 'Comparative Advantage and Economic Geography', *CEPR Discussion Paper*, 2618.

Neumark, D (1999) 'Wage Differentials by Race and Sex: The Roles of Taste Discrimination and Labor Market Information', *Industrial Relations*, 38(3), 414-45.

Nickell, S (1981) 'Biases in Dynamic Models with Fixed Effects', Econometrica, 49(6), 1,417-26.

Nickell, S (1997) 'Unemployment and Labor Market Rigidities: Europe versus North America', *Journal of Economic Perspectives*, 11(3), 55-74.

Nickell, S and Bell, B (1995) "The Collapse in Demand for the Unskilled and Unemployment Across the OECD', Oxford Review of Economic Policy, 11(1), 40-62.

Nickell, S and Bell, B (1996) 'Changes in the Distribution of Wages and Unemployment in OECD Countries', *American Economic Review*, 86(2), 302-8.

Nickell, S and Layard, R (1999) 'Labour Market Institutions and Economic Performance', in (eds) Ashenfelter, O and Card, D, *Handbook of Labor Economics*, vol 3, Amsterdam : North Holland.

Oaxaca R (1973) 'Male-Female Wage Differentials in Urban Labor Markets', *International Economic Review*, 14(3), 693-709.

Oaxaca, R and Ransom, M (1994) 'On Discrimination and the Decomposition of Wage Differentials', *Journal of Econometrics*, 61(1), 5-21.

Owen, G (1999) From Empire to Europe: The Decline and Revival of British Industry Since the Second World War, Harper Collins : London.

Pedroni, P (1999) 'Critical Values for Cointegration Tests in Heterogeneous Panels with Multiple Regressors', Oxford Bulletin of Economics and Statistics, Special Issue, 653-70.

Pesaran, M, Shin, Y, and Smith, R (1998) 'Pooled Mean Group Estimation of Dynamic Heterogeneous Panels', University of Cambridge, mimeo.

Quah, D (1994) 'Exploiting Cross-section Variation for Unit Root Inference in Dynamic Data', *Economics Letters*, 44, 9-19.

Ravn, M and Uhlig, H (2001) 'On Adjusting the HP-Filter for the Frequency of Observations', CEPR Discussion Paper Series, 2858.

Steedman, H and Wagner, K (1989) 'Productivity, Machinery, and Skills: Clothing Manufacture in Britain and Germany', *National Institute Economic Review*, May, 40-57.

Sargan, D (1958) 'The Estimation of Economic Relationships Using Instrumental Variables', *Econometrica*, 26, 393-415.

Swaffield, J (1999) 'Gender, Experience, Motivation, and Wages', *Centre for Economic Performance Working Paper*, 1048, London School of Economics.

Trefler, D (1995) 'The Case of the Missing Trade and Other Mysteries', *American Economic Review*, 85(5), pages 1029-46

Turner, A (2001) Just Capital: The Liberal Economy, Macmillan : London.

Wood, A (1994) North-South Trade, Employment and Inequality: Changing Fortunes in a Skill-Driven World, Oxford: Clarendon Press.

Woodland, A (1982) International Trade and Resource Allocation, North Holland: Amsterdam.

CENTRE FOR ECONOMIC PERFORMANCE Recent Discussion Papers

544	S. Machin A. Manning J. Swaffield	Where the Minimum Wage Bites Hard: the Introduction of the UK National Minimum Wage to a Low Wage Sector
543	R. Belfield D. Marsden	Matchmaking: the Influence of Monitoring Environments on the Effectiveness of Performance Pay Systems
542	C. A. Pissarides	Consumption and Savings With Unemployment Risk: Implications for Optimal Employment Contracts
541	M. Amiti C. A. Pissarides	Trade and Industrial Location with Heterogeneous Labor
540	G. Duranton H. G. Overman	Testing for Localisation Using Micro-Geographic Data
539	D. Metcalf	Unions and Productivity, Financial Performance and Investment: International Evidence
538	F. Collard R. Fonseca R. Mu Z oz	Spanish Unemployment Persistence and the Ladder Effect
537	C. L. Mann E. E. Meade	Home Bias, Transactions Costs, and Prospects for the Euro: A More Detailed Analysis
536	M. Manacorda E. Moretti	Intergenerational Transfers and Household Structure. Why Do Most Italian Youths Live With Their Parents?
535	D. Quah	One Third of the World's Growth and Inequality
534	D. Quah	Matching Demand and Supply in a Weightless Economy: Market Driven Creativity With and Without IPRs
533	R. Dickens A. Manning	Has the National Minimum Wage Reduced UK Wage Inequality?
532	S. Machin A. Manning	The Structure of Wages in What Should be a Competitive Labour Market
531	R. Lydon A. Chevalier	Estimates of the Effect of Wages on Job Satisfaction
530	A. Bryson	The Union Membership Wage Premium: An Analysis Using Propensity Score Matching

529	H. Gray	Family Friendly Working: What a Performance! An Analysis of the Relationship Between the Availability of Family Friendly Policies and Establishment Performance
528	E. Mueller A. Spitz	Managerial Ownership and Firm Performance in German Small and Medium-Sized Enterprises
527	D. Acemoglu J-S. Pischke	Minimum Wages and On-the-Job Training
526	J. Schmitt J. Wadsworth	Give PC's a Chance: Personal Computer Ownership and the Digital Divide in the United States and Great Britain
525	S. Fernie H. Gray	It's a Family Affair: the Effect of Union Recognition and Human Resource Management on the Provision of Equal Opportunities in the UK
524	N. Crafts A. J. Venables	Globalization in History: a Geographical Perspective
523	E. E. Meade D. Nathan Sheets	Regional Influences on US Monetary Policy: Some Implications for Europe
522	D. Quah	Technology Dissemination and Economic Growth: Some Lessons for the New Economy
521	D. Quah	Spatial Agglomeration Dynamics
520	C. A. Pissarides	Company Start-Up Costs and Employment
519	D. T. Mortensen C. A. Pissarides	Taxes, Subsidies and Equilibrium Labor Market Outcomes
518	D. Clark R. Fahr	The Promise of Workplace Training for Non-College Bound Youth: Theory and Evidence from Germany
517	J. Blanden A. Goodman P. Gregg S. Machin	Change in Intergenerational Mobility in Britain
516	A. Chevalier T. K. Viitanen	The Long-Run Labour Market Consequences of Teenage Motherhood in Britain

To order a discussion paper, please contact the Publications Unit Tel 020 7955 7673 Fax 020 7955 7595 Email info@cep.lse.ac.uk Web site http://cep.lse.ac.uk